

"Bodh", BPIT's International Journal of Technology & Management ISSN: 2454-8421, Volume 7, Issue 2, July- Dec 2021, Page 1-10

# Automatic Image Tagging Using Tensor and Gaussian Filter

Tanisha Madan<sup>1</sup>, Tushar Patnaik<sup>2</sup>, Deepali Virmani<sup>3</sup>

<sup>1</sup>Department of Computer Science and Engineering, Bhagwan Parshuram Institute of Technology, Delhi, India, <sup>2</sup>Education & Technology department, Centre for Development of Advanced Computing, Delhi, India <sup>3</sup> Artificial Intelligence and Data Sciences Department, VIPS-TC College of Engineering, Delhi, India <sup>1</sup>tanishamadan@gmail.com, <sup>2</sup>tusharpatnaik@cdac.in, <sup>3</sup>deepalivirmani@gmail.com

Abstract- As automatic image tagging is important in commercial and research, there is a wide gap in visual representation and text generated by image. There is need to assign better quality tag to an image and remove the need of manual tagging. Gaussian filter is used to improve the quality of low-level visual features of the image and to minimize the semantic gap. All images are converted into tensors to form a group of similar features. 3-level tucker decomposition is then processed on tensors to find the better matching of the context group. Tensor formation and context estimation is used in the proposed approach to minimize the semantic gap. Sometimes irrelevant tags are assigned to the image and sometimes tags are not retrieved so this problem had been reduced through three level tensor decomposition and Gaussian blurring filter. The proposed algorithm is tested on the corel-10K dataset.

Keywords--- Tensor, Gaussian, Decomposition, Corel-10K

#### I. INTRODUCTION

In the last two decades a large amount of research has been carried out in automatic images tagging. Generally, the focused research part in this area is content based image retrieval system. While the current research shows there is a large semantic gap between image semantics understandable by human and content based image retrieval [1]. Distance between features that are visual in the image and the tag generated for the image is known as semantic gap [2]. Therefore, a large amount of research has been performed to minimize the semantic gap between tag generated and visual representation of image. In this paper the context of an image based algorithm is used to mend this gap. The information provided in the context can only complete the gap between the textual description and content of the image [2]. Context estimation is an essential task to incorporate the information for a better image annotation process. The context generated from the image itself with the help of visual features present in the image.

In this paper Gaussian filter based feature-independent and unsupervised context estimation is proposed for better classification and results are compared with the classical feature-independent and unsupervised context estimation method. Gaussian filters are used in this paper to improve the quality of visual representation of images and to minimize semantic gap. All the context images are converted into tensors to form a group of similar features; image.3 level tucker decompositions are then processed on tensors to find the better matching of the context group. The proposed algorithm is tested on the corel-10K dataset. Precision, accuracy, and recall are calculated for results and analysis purposes. The paper is organized as follows. Section1 deals with introduction. Section 2 deals with the related work. Problem formulation is described in the Section 3, Section 4 and its subsection deals with the estimation of the context information. Mathematical model for context estimation comes in Section 5. Section 6 deals with results and discussion. Section 7 deals with conclusion and future scope. Section 8 contains the reference part.

#### II. RELATED WORK

Automatic generation of image annotations has been studied for many years with the increasing popularity of social media. Several social media based approach [3,4, 5] have and been developed and proved its role in traditional applications as well as for personal needs. There are many methods in the past that are being developed for tagging but major drawback is that they are not giving appropriate tag to an image i.e. lacking the semantic gap. Machine translation of relevance models has been adapted for automatic image annotation [6-9]. The joint probability of the images with visual representation and textual description is modelled. These models are used for building a classifier that is used for tagging upcoming images. The tag refinement approach is proposed by which are based on employing random walk on a pair-wise graph, where the mined relations between tags are represented by the edge of the graph [10]. The authors of the research paper [11] have proposed metric learning based weight factor assignment to the neighbour images.

Auxiliary information can also be used with images to produce image tag. Usually, these models work with the images having news datasets. All the images available in the dataset must accompany with some news articles. In order to reduce the semantic gap between textual description and visual representation, auxiliary information based context estimation is used. The context estimation of all images must be performed for better accuracy. The video analysis tensors splitting based technique have been found suitable to recognize goals such as motion detection and action recognition. Automatic image tagging can be built using DenseNet, an advanced deep learning model [13].

AICRL model consist of one encoder and decoder. Encoder is built with ResNet 50 and decoder is built with LSTM [14]. CNN-LSTM is also used to recognize and generating the tag of images [15]. Neural network approaches are best in determining the tag of images automatically [16, 17]. Bengali tags can also be generated to tag the images [18]. AlexNet and GoogLeNet are also built for images tagging [19]. Automatic annotation can also be done through mask RCNN and object can be detected through AWS [20]. The tensor based approach is used to better combine the similar context groups and 3-level tucker decomposition is used to evaluate the better correlated matrix. The variance matrix is used to better predict the tag for the testing images.

#### III. PROBLEM FORMULATION

The objective of this paper is to incorporate proper textual tag information for the process of automatic image annotation. For better accuracy estimation of the context information is used to incorporate the exact textual tag. Corel-10K[12] dataset is used for the training and testing purposes. First training folder contains set of training images with textual description. Let there are A training images having AxR vocabulary sets having the textual description of the image. All the training images will be filtered and form a context group. The context group must be form in such a way that all the images in one group must have some relation with the images in another group. The automatic image annotation will be performed on the testing images having M samples. The testing images will be annotated on the basis of trained samples. Estimation of the context information and context group formation are defined as in [2]. The flowchart of the proposed approach is as in fig1.



Fig 1: Proposed approach

The flowchart of proposed approach uses the major steps as:

- 1) Context group formation
- 2) Tensor Formation and Decomposition
- 3) Context Estimation
- 4) Context Based Automatic Image Annotation

### IV. CONTEXT INFORMATION

Three step context estimation processes are proposed in this paper. In the first step groups having a limited number of training data are formed such that there must be some relation between images in different groups. The groups are then converted into tensors and then 3-level tucker decomposition is performed on the tensors to get better features. In the last step the textual description matrix called the label matrix is formed and assigned a number corresponding to the textual description. Example: Suppose textual description of an image are grass, water and animal then for ease of programming purposes grass will be assigned as 1, water as 2, and animal as 3.

#### A. Context Groups

In the first step context groups are formed that have high visual similarity between the images. The method being proposed requires two things:

1) The images should have high context similarity for better prediction for testing images.

2) Different groups must have some visual similarity for better learning and prediction purposes. The textual description of all the groups is available in the label matrix as discussed in the above section and must be highly correlated with the visual representation. Since each group contains N training images therefore tf- idf represents the X, denoted by t which is a vector having length A.

If  $A_{freq}$  is the frequency of ath word appearance in the textual description of the image X and  $A_{total}$  is the cumulative frequency of nth word in the dataset, then the value of nth sample in the vector t will be

$$h_A = \frac{A_{total}}{A_{freq}} \tag{1}$$

The images are clustered on the basis of cosine similarity tf-idf vectors. This process ensures that visual similar image will come into the same context group. All the groups will be divided into two parts 1. General features based grouping and 2. Distinctive features based grouping. Each image will contain either general features or distinctive features. In the example, if the word 'water' occurs frequently in the textual description of all the images then it will be the 'general' vocabulary of the dataset. On the other hand, if 'animal' is textual description of few images then it will be 'distinct' vocabulary of the dataset.



Fig 2: Context group example based on visual similarity in the textual description

#### B. Tensor Formation and Decomposition

In the first step one context tensor  $S \in \mathbb{R}^{X \times m \times n}$  is constructed for every groups of image as in the Fig. 2. The images are resize to a fixed dimension (height and width), gray-scale conversion, and passed through the Gaussian filter. Post filtering they combined to form a tensor where X, Y, and Z are width, height and number of images.

» t	
t is a tensor of size 60 x 60 x 5	
t(:,:,1) =	
Columns 1 through 20	
43 79 90 91 91 90 91 92 93 94 94 94 95 96 97 97 96 96	96 96
73 116 125 126 126 126 127 129 130 130 130 131 133 133 133 133 133	134 134
85 125 129 130 132 132 133 135 137 136 136 136 136 137 137 137 136 138	136 136
89 127 129 129 131 132 133 134 136 136 135 135 135 136 136 135 134 133	133 134
90 128 131 131 132 133 135 135 135 134 135 135 134 134 134 134 134	134 134
91 128 131 131 131 131 133 135 135 134 134 135 134 133 132 133 133 134	133 133
91 126 128 129 128 130 132 133 132 131 131 130 129 129 129 129 130	130 130
89 124 126 126 126 129 131 130 129 129 129 129 128 128 126 126 126 127	128 128
90 125 128 127 128 129 130 130 129 130 130 130 131 130 128 127 127 128	129 129
90 126 130 128 129 130 129 127 127 128 127 127 128 129 129 128 128 128	129 128
89 125 130 127 127 127 125 123 123 124 121 120 123 127 128 127 127 127	127 126
85 120 124 123 122 121 120 119 120 120 118 118 119 122 122 123 124 124	125 124
	120 120
85 115 117 118 116 116 115 115 114 115 115 115 115 116 113 113 112 112	115 115
	110 118
84 114 11/ 116 116 11/ 119 119 119 118 116 115 115 115 117 117 117	11/ 119

Fig 3: Tensor of context group

The tensor  $S \in R^{X \times Y \times Z}$  are decomposed into smaller core tensor S and the matrices A, B and C such that

$$S \approx S \times_1 A \times_2 B \times_3 C = \sum_{i=1}^X \sum_{j=1}^m \sum_{k=1}^n g_{ijk} a_i b_j c_k$$
(2)

where  $A \in R^{X \times LEV}$ ,  $B \in R^{m \times LEV}$ , and  $C \in R^{n \times LEV}$  are the orthogonal matrices,

 $S \in R^{LEV \times LEV \times LEV}$  is the core tensor and  $LEV \le min(X, Y, Z)$ .

The  $\times_i$  operator denotes the tensor namely,

$$\alpha = \beta \times_i \gamma \ll \gg \alpha_{jk} = \sum_{i=1}^N \beta_{ijk}$$
(3)

where *A*, *B*, and *C* are the matrix having dimensions  $X \times LEV$ ,  $M \times LEV$ , and  $N \times LEV$ . The *LEV* is the rank of tucker decomposition. In this paper *LEV* is taken as 3. *X*, *Y*, and *Z* are the height, width, and the size of the context group. The matrices *A*, *B* and *C* are the similarity/dissimilarity of one image to its neighbour images. Since all the images belong to the same category, it will find a high similarity between all images. The matrix *C* is the compact signature of the context group.

#### C. Context Estimation

In this process the context for the image quantifies for their matching with different signature contexts.  $X_0$  denotes the test image. Since there is no textual description available for the testing image, the variance in the elements of data in context signature matrix C will be very minimal because of the highly correlated visual appearance. If any foreign image insert into the context group, then the variance of that image in the context signature matrixC will be easily rejected after the tucker decomposition process. The divergence in variance will be directly proportional to the dissimilarity in the image from the context group. The image which needs to be tagged is inserted at the location L in a tensor t by swapping the images kept at that location.

Now the matrix C' will be computed using tucker decomposition method. |C - C'| is used to measure the test image association with the context of context group into the tensor *T*. The distribution of conditional probability of the test image for every context group is calculated as:

$$W(H|X_0) = \frac{e^{(-(C'-C)^t r^{-1}(C'-C))}}{\sqrt{2\pi|r|}}$$
(4)

where  $\Gamma$  is the covariance matrix,  $X_0$  is the test image.

When the context groups are formed, the textual descriptions are given weight on the basis of their frequency. It is given less weight if it occurs very frequently and vice versa. Each test image  $X_0$  is provided with the same conditional probability as the context group.

#### V. MATHEMATICAL MODEL FOR CONTEXT BASED AUTOMATIC IMAGE ANNOTATION

In Suppose there are n number of visual units such as  $V_n$  in an image and m number of textual description such as  $H_m$ . Let there are *CC* number of context categories where  $T \in CC$  corresponds to one context group. Each training image will belong from the one of these T groups. By picking a context group with conditional probability over test image  $X_0$  i.e.  $W(H|X_0)$ . By selecting a training image  $X_t$  within the training set *TS* with the probability  $W(H|X_h)$ 

for i=1,2,....,n

2.1 Pick a visual unit  $V_i$  having conditional probability  $W_R(.|X_h)$ 

For j=1,2,....,m

2.2 By selecting a word  $h_i$  from conditional probability  $W_T(.|X_h)$ 

The main aim of the proposed approach is to enhance probability metric V and T over the training image  $X_h$  $P(X_h) = \sum_{H \in CC} P(H|X_h) \sum_{X_h \in HS} P(X_h|H) \prod_{i \in m} w_T(H_i|X_h) \prod_{i \in n} W_R(vI/X_t)$ 

$$(5)$$

The  $W_H(H_j|X_t)$  (Bernoulli distribution) is defined as:

$$W_H(H_j|X_h) = \frac{\mu \delta_{H_j} + N_{H_j}}{\mu + N_H}$$
(6)

where,  $A_{H_i}$  is the members of T with word  $T_j$  in their description

 $A_{H_i}$  is members of  $T_j$ 

 $\delta_{H_i}$  is set to be 1 if description of the image  $X_t$  has word  $T_j$  in it  $\mu$  is empirically selected constant

 $W_R(V_i|X_t)$  is the density estimate to generate the visual unit  $V_i$  for the training image  $X_t$ . Gaussian kernel is employed for this density estimate. Suppose if the visual units of the training image  $X_t$  are  $\{VT_1, VT_2, \dots, VT_n\}$  then

$$(V_i|X_h) = \frac{e^{(-(V_i - VH_n)^T (\sum V_i - VH_n)^{-1})}}{\sqrt{2\pi|\Sigma|}}$$
(7)

where  $\sum$  is the covariance matrix.

#### VI. RESULTS AND DISCUSSION

The proposed algorithm is tested on the Corel-10k dataset. In this paper results are compared without using filters and with using filters. In both the cases the tucker decomposition level is taken as 3. To check the effectiveness of the algorithm the comparison has been made between precision, recall, and accuracy. Since the dataset contains 10k images, the complete dataset are divided into small context groups for easy and fast analysis purposes. We have also checked the efficiency of the algorithm by taking different percentage combinations of training and testing data i.e. 70% and 30%, 50% and 50% etc.

Group's name (Training images- Testing images)	No. of correct and retrieved tags in proposed approach	No. of correct and retrieved tags in base approach	Total tags that should be correct and retrieved			
New 17(70-30)	59	45	60			
New 19(70-30)	51	44	60			
New 21(50-50)	101	83	113			
New 12(70-30)	57	56	60			
New 18(50-50)	92	86	100			
New 32(50-50)	106	94	131			
New 23(5-5)	11	11	12			
New 24(5-5)	10	10	10			

#### Table I: Results analysis of different context groups

©"BBIJTM July-Dec. 2021", All Rights ReservedPage 6

The results thus obtained are as in the table I-IV.

- Some tags are correct and retrieved
- Some tags are incorrect and retrieved
- Some tags are not retrieved giving some random value.
- Precision: Precision is calculated as a fraction of relevant tags among retrieved tags as in equation (8).
   Precision P=D/E (8)

where D is the number of relevant images retrieved whereas E is the total number of images retrieved.

Group name(Training images-Testing images)	Proposed approach Precision in %	<b>Base approach Precision in %</b>
New 17(70-30)	98.3333	83.3333
New 19(70-30)	86.4407	81.4815
New 21(50-50)	90.9910	74.778
New 12(50-50)	98.2759	98.2456
New 18(50-50)	92.9293	88.6598
New 32(50-50)	79.6992	71.557

#### Table II: Precision table for both the cases

Recall: Recall is calculated as a fraction of total relevant tags that are retrieved equation (9).
 Recall=D/F (9)

where D is the number of relevant images retrieved. F is the number of images that are relevant in the dataset.

As it can be seen from Table I to IV that all the analysis parameters such as precision, recall, and accuracy have been improved many fold in the proposed approach as compared to the base approach.

Group 'name (Training images-testing images)	Proposed approach recall in %	Base approach Recall in %
New 17(70-30)	100	90
New 19(70-30)	98.0769	95.6522
New 21(50-50)	98.0583	90.2714
New 12(50-50)	96.6102	94.9153
New 18(50-50)	98.9247	97.7273
New 32(50-50)	85.4839	71.557

#### Table III: Recall table for both the cases

Accuracy is calculated as total no of correct observation divided by total no of observations.

Table IV: Accuracy table for both the cases

Group's name(Training images- Testing images)	Proposed approach accuracy in%	Base approach accuracy in%
New 17(70-30)	88.2602	74.9217
New 19(70-30)	74.6538	70.4888
New 21(50-50)	78.1253	72.8023
New 12(50-50)	87.3810	84.7413
New 18(50-50)	83.9296	81.4908
New 32(50-50)	71.9161	64.4773

The model proposed by author Tariq et.al [2] has used single level tucker decomposition while in this paper three level tucker decomposition is used to model the base case. Even in the base case the results are better as compared to the results obtained by Tariq et. al [2]. Since all the images contain Gaussian noise by default therefore adopting the Gaussian filtering technique improves the results.

approach resulta	
accuracy =	
100.0000 94.7400 100.0000 64.8800 100.0000 92.1800 0 100.0000 71.0100 100.0000 55.8900	96.2800
totslaoduracy =	
81.2863	
base paper results	
accuracy =	
100.0000 56.6700 100.0000 79.8900 100.0000 98.4900 0 100.0000 77.0600 100.0000 48.6200	71.5100
cotalacouracy =	
80.1367	

Fig 4: Snapshot of results obtained in MATLAB

The graphical representation of the above table can be seen in Fig. 5 to Fig. 7. MATLAB software is used for the simulation and analysis purposes. The results obtained during simulation are as in Fig. 4.



Fig 5: Precision graph for different context groups of dataset



Fig 6: Recall graph for different context groups of dataset



Fig 7: Accuracy graph for different context groups of dataset

## VII. CONCLUSION AND FUTURE SCOPE

Gaussian filtering based novel context estimation and tensor decomposition system is proposed. Tensor formation and context estimation is used in this research paper to minimize the semantic gap while the tensor decomposition

is used to find the best correlation between the context group images. Due to minimization of semantic gap the accuracy improves significantly.3-level tucker decomposition is adopted to model the framework for better correlation among the context groups. The results are compared between filtered context groups and unfiltered context groups. The evidence of the effectiveness of the proposed algorithm can be seen from the results and discussion section. In future, deep learning techniques can be used for better accuracy as well as to reduce the searching time.

#### REFERENCES

- [1].BahramiS, Abadeh M. S, Automatic Image Annotation Using an Evolutionary Algorithm (IAGA),7th International Symposium on Telecommunication, pp. 320–325, 2014.
- [2]. Tariq A, Foroosh H, Feature-Independent Context Estimation for Automatic Image Annotation, IEEE, pp.1958-1965, 2015
- [3].S. A. Zhu, C.-W. Ngo, and Y.-G. Jiang, "Sampling and ontologically pooling web images for visual concept learning," IEEE Trans. on MM, vol. 14, no. 4, pp. 1068–1078, 2012.
- [4].X.-R. Li, C. G. M. Snoek, and M. Worring, "Learning social tag relevance by neighbor voting," IEEE Trans. on MM, vol. 11, no. 7, pp. 1310–1322, 2009.
- [5].M. Guillaumin, T. Mensink, J. J. Verbeek, and C. Schmid, "TagProp: Discriminative metric learning in nearest neighbor models for image auto-annotation," in ICCV, 2009.
- [6].J. Jeon, V. Lavrenko, and R. Manmatha. Automatic image annotation and retrieval using cross-media relevance models. In Proceedings of the 26th annual international ACM SIGIR conference on Research and development in information retrieval, pp.119-126,2003.
- [7].V. Lavrenko, R. Manmatha, and J. Jeon. A model for learning the semantics of pictures. In Advances in neural information processing systems,2003.
- [8].S. Feng, R. Manmatha, and V. Lavrenko. Multiple bernoulli relevance models for image and video annotation. In IEEE Computer Society Conference on Computer Vision and Pattern Recognition, pp. 1-8,2004.
- [9].S. Moran and V. Lavrenko. Optimal tag sets for automatic image annotation. In Proceedings of the British Machine Vision Conference, pp. 1-11, 2011.
- [10]. D. Liu, X.-S. Hua, L.-J. Yang, M. Wang, and H.-J. Zhang, "Tag ranking," in WWW,pp. 351-360, 2009.
- [11]. A.Rae,B.Sigurbj"ornsson,andR.-V.Zwol, "Improving tag recommendation using social networks," in RIAO, 2010.
- [12]. http://www.ci.gxnu.edu.cn/cbir/Dataset.aspx
- [13]. Tran, T.-H.; Tran, X.-H.; Nguyen, V.-T.; Nguyen-An, K. Building an Automatic Image Tagger with DenseNet and Transfer Learning; IEEE: Piscataway, NJ, USA, pp. 34–41,2019
- [14]. Chu, Y., et al.: Automatic image captioning based on ResNet50 and LSTM with soft attention. In: Wireless Communications and Mobile Computing, 2020
- [15].R. Subash November 2019 Journal of Physics Conference Series 1362:012096 : Automatic Image Captioning Using Convolution Neural Networks and LSTM.
- [16]. Bai, S., An, S.: A Survey on automatic image caption generation. Neurocomputing 311, 291–304 (2018)
- [17]. Tanti, M., Gatt, A., Camilleri, K.: What is the Role of Recurrent Neural Networks (rnns) in an Image Caption Generator? in: proceedings of the 10th International Conference on Natura l Language Generation, pp. 51–60 (2017).
- [18]. Kamal, A.h., Jishan, M.a., Mansoor, N.: Textmage: The Automated Bangla Caption Generator Based on Deep Learning. in: 2020 International Conference on Decision aid Sciences and Application (DASA), pp. 822–826. IEEE (2020)
- [19]. Teera Siriteerakul, Kunlabut Suriyakanon, Sofia Sarideh, (2018) "Automatic Restaurant Image Tagging", International Journal of Electrical, Electronics and Data Communication (IJEEDC), pp. 1-4, volume-6, issue-4
- [20]. Marielet Guillermo, Robert Kerwin Billones, Argel Bandala, Ryan Rhay Vicerra, Edwin Sybingco, ElmerP. Dadios, Alexis Fillone, "Implementation of Automated Annotation through Mask RCNN Object

Detection Model in Cvat using AWS ec2 instance", Region 10 Conference (TENCON) 2020 IEEE, pp. 708-713, 2020.



"Bodh", BPIT's International Journal of Technology & Management ISSN: 2454-8421, Volume 7, Issue 2, July- Dec 2021, Page 11-18

Dynamic Operational Time Traffic Light System Drishti Agarwal<sup>1</sup>, Preeti Nagrath<sup>2</sup>, Rachna Jain<sup>3</sup>, Narina Thakur<sup>4</sup>

<sup>1, 2</sup> Department of Computer Science and Engineering Bharati Vidyapeeth's College of Engineering

New Delhi, India drishtis.agarwal@gmail.com

<sup>3</sup> Department of Information Technology, Bhagwan Parshuram Institute of Technology New Delhi, India 4 Department of Computer Science and Engineering Bhagwan Parshuram Institute of Technology New Delhi, India 1drishtis.agarwal@gmail.com, <sup>2</sup>preetinagrath1@gmail.com, <sup>3</sup>rachnabtec@gmail.com, <sup>4</sup>narinat@gmail.com

Abstract - The process of detecting, tracking, and counting moving vehicles is crucial for traffic flow monitoring, control, and planning. In comparison to other solutions, video-applicable solutions are simple to set up and do not obstruct traffic flow. This paper provides a video-based system with the help of advanced methods implemented in python language and OpenCV computer vision kits, indicating that the outlined approach can effectively perform detection, tracking, counting of moving vehicles, and calculate the operational time for traffic lights by examining the traffic flow video series recorded from a video-based source (camera).

Keywords--- Tensor, OpenCV, traffic, operational time, detect, vehicle, count.

#### I. INTRODUCTION

With the growing number of automobiles, expressways, highways, and roadways are becoming increasingly congested. Smart Transportation Systems are being developed around the world to gather, process, and organize data from diverse sources to improve the efficiency and reliability of transportation flows and render them increasingly effective, smarter, and safe. As a result, the process of detecting, following, and counting the number of passing vehicles is becoming increasingly crucial for traffic control. Infrared detectors, inductive loop detectors, radar detectors, and video-based systems are used in the traditional method of vehicle detection. Surveillance cameras are highly influenced by the external environment, such as harsh weather conditions, shadows, sun- light, and so on, therefore video-implemented systems have an advantage. Moreover, video-based surveillance systems can provide many benefits such as easy setup, conveniently modified, do not disturb the traffic flow etc. As a result, they have attracted researchers all over the globe [1]. Road congestion is a significant concern that impacts many regions throughout the world. Numerous factors contribute to the heavy traffic situation.

The population of citizens settling in cities has increased significantly, resulting in a surge in the number of automobiles. Roadway infrastructure, on the other hand, has developed slowly and is now insufficient. As a result, there is a mismatch between the number of cars and the number of roads, leading to traffic, particularly in large cities. Inadequacies in public transit networks contribute to the same issue. Another reason is that the traffic management system is flawed and does not provide real-time traffic updates. If the traffic issue is not addressed, it appears to become worse in the future. Intelligent transportation systems are adaptive, intending to increase vehicle flow (the number of vehicles passing in a given time-frame) and decrease the average wait time interval for all cars in a system-controlled area. As a result, the purpose of this model is to build and implement a traffic control system based on real-time data sources using appropriate and favorable algorithms. The real-time vehicle data is acquired from traffic cameras installed at a four-way intersection. For this purpose, we have used a powerful tool OpenCV that is Open Source Computer Vision Library in Python. By using OpenCV, we were able to develop real-time computer vision applications. Road congestion issues have also been solved using computer vision. Take, for example, a video stream of the road that may be analyzed to identify and measure the total number of vehicles on the road. Computer vision can also be used to calculate additional information such as a vehicle's speed or traffic density. This will also benefit two categories of people: commuters and traffic management personnel. Moreover, if commuters are already aware of real-time traffic data, one can utilize it to determine the optimal mode of transportation and avoid traffic congestion. Furthermore, traffic authorities can include information on traffic density flow into their traffic control systems, resulting in improved traffic control. On realtime traffic film, numerous approaches for vehicle detection and counting have been developed. We must first detect the blobs in the movie before counting them. This is done by a series of processing on every frame of the video such that the blobs can be detected precisely.

The whole process of detection involves 3 major steps: Background Subtraction followed by vehicle detection and then finally counting them. Firstly, we take the grayscale background image of the road and calculate the grayscale difference between the current grayscale frame and the average of the scene so that the background is masked out and the blobs are detected. Then the frame is applied to various filters such as Gaussian Blur to smoothen the

image so that detecting whole blobs as one unit becomes easy. Secondly, by performing a binarization process to obtain a foreground area, and then with the help of morphological operations to eliminate noise and shadow. Once the blobs are detected the next thing is to count them in such a way that no blobs are counted more than once. It is ensured by a calculating distance of blobs in consecutive frames and if it turns out to be less than a threshold value it's considered as the same vehicle. Once all the vehicles in the lane are calculated for each road in the junction and lastly based on traffic density the operational time gets calculated for each traffic light. The paper is organized as follows:

- I. Related Work
- II. Data Collection
- III. Methodology
- IV. Experimental Results
- V. Conclusion and Future scope.

#### II. RELATED WORK

In September 2012, Vivek Tyagi, Senior Member, IEEE with his team published their research. The weightings of the mixture in the accumulated signal vary based on the traffic density conditions [6], in which the identified occupancy is based on the noise from vehicles, which contains numerous noise signals. They collected the spectral components of the roadside acoustic data utilizing Gaussian mixture concepts and modeled the class conditional probability distributions of these feature vectors conditioned on broad traffic density states: jammed, mediumflow, and free-flow traffic (GMMs). The GMM parameters are created by manually labelling 2.5 hours of roadside acoustic data traffic density levels. When 20 to 30 seconds of audio input proof is supplied, a Bayes' model is used to classify traffic density groups based on these dispersions, with approximately 95 percent accuracy. A radial basis function (RBF) kernel classifier based on a support vector machine (SVM) was often used. [7]. CHEN Wenjie, CHEN Lifeng, and CHEN Zhanglong of Fudan University [8] submitted their work on developing a dynamic traffic signal model based on traffic detection utilising sensors. The traffic density was calculated using a range of sensors, thus it was a traditional and costly technique to deal with the problem. In 2007, Erhan Bas, published a detection technique for counting the volume of traffic in surveillance video. They proposed an innovative transportation video analysis method that takes into account the architecture of the scene and employs adaptive bounding box dimensions to identify and detect vehicles based on their approximate distance from the camera. Throughout the rest of the study, they recommend algorithms for vehicle detection and tracking and present results obtained by applying these algorithms to various video footage gathered at various time periods at one place in Istanbul [9].

Luis Unzueta and colleagues submitted a paper in 2012. They developed an adaptive multi-cue segmentation method which distinguishes between pixels associated with mobile and stopped cars. First, an adaptive threshold is applied to a mixture of brightness and chrominance discrepancy charts between the learned backdrops and the present frame. It tends to add additional information derived from gradient distinctions to aid in the classification of shadowy automobiles of rendered shadows, as well as the removal of roadside headlamp reflections. The segmentation is then approached in two steps, combining the ease of a linear 2-D Kalman filter with the difficulty of a 3-D volume approximation using Markov chain Monte Carlo (MCMC) methodologies. Experiments show that technology can number and categorize objects in real- time with performance comparable to inductive loop detectors in a variety of environments [10].

#### III. DATA COLLECTION

The data collected can be from any video source such as webcam, CCTV-cam, mobile cam etc. The model can work on live video feed from the traffic junction. For demonstration purposes the video data is collected from: https://s3-eu-west-1.amazonaws.com/jamcams.tfl.gov.uk some video samples are as in Fig.1.



Fig.1. (a) Day time CCTV footage (b) Night time CCTV footage

#### **METHODOLOGY** IV.

This Section is all about the detection and counting of vehicles in the video footage. This section is further divided into three subsections giving the details of each step.

#### **Foreground Extraction Or Background Subtraction** A.

Foreground extraction is the process using which we can extract the foreground components of any frame. In particular, a background image of the road, which contains no vehicle, and the current frame in the video are converted from color (RGB) to HSV image. The recorded frame's valuation is then subjected to a non-linear bilateral filter. It converts each pixel's intensity to grayscale by replacing it with a weighted average of intensity values from neighbouring pixels. The gray concentration of the backdrop picture is then deducted from that of the current frame for each pixel (x, y). The absolute result is saved in the same position in another image, which is referred to as a different image.

#### B. **Vehicle Counting**

Once the traffic density is counted using OpenCV, the final thing is to calculate the operation traffic light duration and compare the results with the static duration of traffic lights. Before jumping to calculation part, we made some assumptions:

- No. of Vehicles in each road is a random value between 25-200.
- Rate at which vehicle leaves the traffic signal: Heavy Traffic: 25 vehicle/second, Medium Traffic: 20 vehicle/second, Low Traffic: 15 vehicle/second
- Static timed traffic signal duration is 6 seconds. For each cycle, the traffic signal with highest operational time will glow first then next turn will be in clockwise direction.



Fig.2. (a) Image of the road with no vehicles (b) After applying Bilateral Filter (c) Foreground Extraction (d) Foreground Extraction after Gaussian blur

After each Fig. 3 (a) Mean thresholding on starting frames (b) Mean thresholding on frames after sometime (c) Merging blobs after removal of noise and hole filling (d) Setting up region of interest complete cycle, the traffic signals will reset themselves and start operating in the similar fashion before.



Fig.4. Final output screen counting and tracking each vehicle in Region of Interest (ROI).

#### C. Time Calculation

For the operation of traffic lights, we need to define some variables to store the required information needed for proper working of traffic lights. Using the python library numpy, we will create an array by the name of 'total' which will contain the total number of vehicles present in each lane for the respective traffic light, then we will create another array by the name of 'rate' which will contain the rate of vehicles arriving per second, after that another array by the name of 'dept' will contain the departure rate for every lane and two more arrays will be created by the name of 'old vehicle' contains the total number of vehicles remaining after each second during operational time of each traffic light . Another variable 'active' is defined which will tell us the highest operational time required by a particular traffic light. Some functions we have to define in order to implement the working of traffic lights in an orderly manner.

- 1. **time cal** : It will calculate the operational time for each traffic light, that is the time interval till which each traffic light will be green during their respective turn and store it in an array by the name of 'time'.
- 2. **time dec:** Responsible for decreasing the operational time, that is for how much duration a particular traffic light will be switched to 'green' signal by unity.

- 3. **car dec:** It will calculate the number of vehicles re- maining after each second for the respective traffic light which had been set to indicate 'green' signal.
- 4. **light sat:** It is used to change the signal of a particular traffic light according to its operation.
- 5. **light operation:** It is used to operate all the traffic lights in an appropriate manner, that is by diverting the traffic flow in descending order. In order to maintain the stability along all the paths. This function is also responsible for calling the above functions by fulfilling their conditions

By implementing nested for loop, in order to call the light operation function and by updating the value of 'active' when operational time of one traffic light has been completed by deleting the highest operational time present in 'time' till all elements are used. The execution of all the traffic lights will take place in such a way that traffic lights with the highest traffic density will be diverted first, then second highest traffic lights will be diverted and so on. Till all traffic lights have diverted the traffic flow and the cycle resets itself to be implemented again.

#### V. EXPERIMENTAL RESULTS

To achieve the desired result from the proposed model, first we have to take into consideration the required information for its implementation. As a result, we have taken into consideration a set of assumed data which is as follows. Initially, all the traffic lights display's 'RED' lights, from the beginning, we have taken the maximum number of vehicles in a frame to be 200 and minimum number of vehicles in a frame to be 200 and minimum number of vehicles in a frame to be 20. In order to have efficient detection of vehicles, we divided the traffic density into three parts namely Low, Medium and Heavy.

The respective number of vehicles required to differentiate between each type of traffic is as follows, Greater than 140 (Heavy), Less than 140 but greater than 70 (Medium) and Less than 70 (Low). The departure rate that is number of vehicles entering in a particular frame per second for each type of traffic is assumed to be 25 vehicles/sec for Heavy traffic, 20 vehicles/sec for Medium traffic and 15 vehicles/sec for Low traffic. Hence, the operational time in dynamic system is calculated with the help of above information.



Fig.5. (a) Comparison between static and dynamic operational time (b) Comparison between total numbers of vehicles left out after implementation of static and dynamic operational time system

Finally, in order to compare our results with the static system, we have assumed the operational time (Time interval till which 'GREEN' light is displayed) of each traffic light to be 6 seconds and departure rate is same as in case of dynamic system. Hence, the experimental results displayed above are calculated, as in Fig 5(a) and Fig5 (b).

#### VI. CONCLUSION AND FUTURE SCOPE

In this paper, we proposed a real time object detection approach for calculation of vehicles, and then estimated the required time for the traffic flow. After that, we used the estimated time and operated the traffic lights accordingly depending upon the traffic density in a sequential manner. Image Processing played an important role in extracting the information from images. After the processing approach, we implemented the object detection techniques using OpenCV like frame differencing to notice the changes in coordinates of the vehicles under consideration in under consecutive frames and using thresholding technique to separate the vehicles and the unwanted objects from the images in order to increase the efficiency of the detection of vehicles and thus increased the accuracy for counting process of vehicles and as explained in detail in section 4. After completion of the detection process, we counted the total number of vehicles in our frame using user defined functions in python. Once the counting was completed, the operation of traffic lights was handled by the user defined functions and thus traffic lights were operated accordingly depending upon the live traffic density in an orderly manner. In the present day scenario, implementation of such a model requires a lot of different types of physical components like Arduino, camera, light bulbs, storage unit, durable structure build etc. Moreover, a GPU is also required for processing of images in real time for implementation of object detection process. A storage unit is also required, since the count of vehicles has to be stored in order to calculate the implementation time for each traffic light. An interconnected wired system has to be implemented in order to operate each traffic light accordingly. A reliable power source is also required for reasonable and error free working of the system. The requirement of a power source can be fulfilled in one way by using a solar cell and a rechargeable battery installed along the system which can be used during day as well as at night. At last, a powerful computer system is also required in order to handle execution of all the processes in a reasonable time frame so that the operation of traffic lights can be in sync with each other and thus the proposed system can work efficiently and at a reasonable fast pace in order to cope up the increasing traffic density without dealing with any problem. By fulfilling all the stated requirements, the proposed system can work efficiently in any environment and thus eliminating the problem of unsupervised traffic flow in the world.

#### REFERENCES

- Li, D., Liang, B., Zhang, W. (2014, April). Real-time moving vehicle detection, tracking, and counting system implemented with OpenCV. In 2014 4th IEEE International Conference on Information Science and Technology (pp. 631-634). IEEE.
- [2] Seenouvong, N., Watchareeruetai, U., Nuthong, C., Khongsomboon, K., Ohnishi, N. (2016). A computer vision based vehicle detection and counting system. 2016 8th International Conference
- [3] Y. Liu, G. Li, S. Hu and T. Ye, "Real-time detection of traffic flow com- bining virtual detection-line and contour feature," Proceedings of 2011 International Conference on Transportation, Mechanical, and Electrical Engineering (TMEE), pp. 408–413, December 2011.
- [4] S. A. Amman and M. Das, "An efficient technique for modeling and synthesis of automotive engine sounds," IEEE Trans. Ind. Electron., vol. 48, no. 1, pp. 225–234, Feb. 2001.
- [5] Tyagi, V., Kalyanaraman, S., Krishnapuram, R. (2012). Vehicular traffic density state estimation based on cumulative road acoustics. IEEE Transactions on Intelligent Transportation Systems, 13(3), 1156-1166.
- [6] Chen Wenjie, Chen Lifeng, Chen Zhanglong and Tu Shiliang, "A realtime dynamic traffic control system based on wireless sen- sor network," 2005 International Conference on Parallel Process- ing Workshops (ICPPW'05), Oslo, Norway, 2005, pp. 258-264, doi: 10.1109/ICPPW.2005.16.
- [7] E. Bas, A. M. Tekalp and F. S. Salman, "Automatic Vehicle Counting from Video for Traffic Flow Analysis," 2007 IEEE In- telligent Vehicles Symposium, Istanbul, 2007, pp. 392-397, doi: 10.1109/IVS.2007.4290146.
- [8] L. Unzueta, M. Nieto, A. Cortes, J. Barandiaran, O. Otaegui and
- [9] P. Sanchez, "Adaptive Multicue Background Subtraction for Robust Vehicle Counting and Classification," in IEEE Transactions on Intelligent Transportation Systems, vol. 13, no. 2, pp. 527-540, June 2012, doi: 10.1109/TITS.2011.2174358
- [10] Betke, M., Haritaoglu, E., Davis, L. S. (2000). Real-time multiple vehicle detection and tracking from a moving vehicle. Machine vision and applications, 12(2), 69-83.
- [11] Kowsari, T., Beauchemin, S. S., Cho, J. (2011, October). Real- time vehicle detection and tracking using stereo vision and multi-view AdaBoost. In 2011 14th International IEEE conference on intelligent transportation systems (ITSC) (pp. 1255-1260). IEEE.
- [12] Jazayeri, A., Cai, H., Zheng, J. Y., Tuceryan, M. (2011). Vehicle detection and tracking in car video based on motion model. IEEE Transactions on Intelligent Transportation Systems, 12(2), 583-595.
- [13] Liu, W., Wen, X., Duan, B., Yuan, H., Wang, N. (2007, June). Rear vehicle detection and tracking for lane change assist. In 2007 IEEE intelligent vehicles symposium (pp. 252-257). IEEE.

- [14] Chen, Y. L., Wu, B. F., Fan, C. J. (2009, October). Real-time vision- based multiple vehicle detection and tracking for nighttime traffic surveillance. In 2009 IEEE International Conference on Systems, Man and Cybernetics (pp. 3352-3358). IEEE.Xiao, J., Cheng, H., Sawhney, H., Han, F. (2010, June). Vehicle detection and tracking in wide field-of-view aerial video. In 2010 IEEE Computer Society Conference on Computer Vision and Pattern Recognition (pp. 679-684). IEEE.
- [15] Bedruz, R. A., Sybingco, E., Bandala, A., Quiros, A. R., Uy, A. C., Dadios, E. (2017, December). Real-time vehicle detection and tracking using a mean-shift based blob analysis and tracking approach. In 2017IEEE 9th International Conference on Humanoid, Nanotechnology, Information Technology, Communication and Control, Environment and Management (HNICEM) (pp. 1-5). IEEE.
- [16] Chellappa, R., Qian, G., Zheng, Q. (2004, May). Vehicle detection and tracking using acoustic and video sensors. In 2004 IEEE International Conference on Acoustics, Speech, and Signal Processing (Vol. 3, pp. iii-793). IEEE.
- [17] Di Stefano, L., Viarani, E. (1999). Vehicle detection and tracking using the block matching algorithm. Proc. of 3rd IMACS/IEEE, 1, 4491-4496.
- [18] Kim, Z., Malik, J. (2003, October). Fast vehicle detection with proba- bilistic feature grouping and its application to vehicle tracking. In null (p. 524). IEEE.
- [19] Caraffi, C., Voj'ı'r, T., Trefny', J., S'ochman, J., Matas, J. (2012, September). A system for real-time detection and tracking of vehicles from a single car-mounted camera. In 2012 15th international IEEE conference on intelligent transportation systems (pp. 975-982). IEEE.
- [20] Huang, L. (2010, March). Real-time multi-vehicle detection and sub-feature based tracking for traffic surveillance systems. In 2010 2nd International Asia Conference on Informatics in Control, Automation and Robotics, CAR 2010, Vol. 2, pp. 324-328, IEEE.



Bodh", BPIT's International Journal of Technology & Management ISSN: 2454-8421, Volume 7, Issue 2, July- Dec 2021, Page 19-25

# A Multi-Hierarchical Clustering on Topic Modeling using Latent Dirichlet Allocation

Anusha Chhabra<sup>1</sup>, Monika Arora<sup>2</sup>

<sup>1</sup>Research Scholar, Delhi Technological University, Delhi, India <sup>2</sup>Department of Information Technology, Bhagwan Parshuram Institute of Technology, Delhi, India,

<sup>1</sup>anusha.chhabra@gmail.com, <sup>2</sup>monikaarora@bpitindia.com

Abstract- With the economic crisis, it is expected that credit and down payment growth will average later on. Credit score growth will be led by spending on the infrastructure while retail credit will display an average growth. Margin demands due to lag effect of quantity cuts between attention quantity on deposits and advances, reduced treasury gains and core fee earnings and improving in conditions for NPAs is likely to put pressure on the main point here of the financial organizations. In the light of above aspects the present document tends to analyze sectoral distribution of credit in Native India Financial industry. The document studies the styles in credit growth with a perspective to project upcoming courses of growth in bank credit.

Key words: Unsupervised learning, K-Means Clustering, MALLET, WEKA

#### I. INTRODUCTION

Topic modeling is the technique utilized in the area of text analysis. It is used for finding the abstract topics occurring in an assortment of documents, which provides us with strategies to arrange, comprehend, and sum up enormous assortments of text based data. It is unique in relation to rule-based text analysis approaches that utilize regular expressions or word reference based keyword-finding methods. It additionally helps in finding hidden topical patterns available across the assortment of data. Topic models have numerous applications in natural processing dialects. Many papers have been published using topic modeling approaches in different subjects like online social media stages, software engineering, Cognitive science and many more. Latent Dirichlet Allocation (LDA) [1], a generative probabilistic topic model is an amazing asset of topic modeling that permits a set of perceptions to be clarified by unobserved groups. The instinct behind LDA is that data archives exhibit numerous topics. Each record shows the topics to various extent; each word in each archive is extracted from one of the topics, where the chosen topic is taken from the per-report dissemination over topics [1], [2].

The unsupervised learning via probabilistic topic model [3] has been successfully developed for document categorization[3] [1], image analysis [4], text segmentation [5], speech recognition [6], information retrieval [7], document summarization [8], [9], and many other applications. Using topic models, latent semantic topics are learned from a bag of words to capture the salient aspects embedded in data collection. If the number of clusters is considered as a number of topics and the probabilities as the extent of cluster participation, then LDA is used as a way of soft-clustering. Cluster analysis is an errand of grouping a set of objects so that objects of the same group are more similar in any sense to one another than to those in other groups. In LDA, the order of archives doesn't make any difference except hyper-parameters which rely upon whether Dirichlet distributions are assumed to be symmetric or asymmetric. For the symmetric distribution considering on alpha value, a high alpha-value is considered means that each corpus is probably going to contain a combination of maximum number of topics, not any single one explicitly [10].

On the other hand, low alpha value puts less such constraints on corpus with the meaning where it is more likely that a corpus contains the combinations of a few or even one topic. Whereas If considering beta values, a higher beta-value represents that each and every topic is likely to have a combination of maximum number of words, not any word explicitly, while a low beta value represents that a topic contains a combination of just a few of the words [1].

Hyper- parameters	Relevance	> values	< values
Alpha	Document-topic density	Documents composed of large number of topics	Documents contain fewer number of topics
Beta	Topic-word density	Topics composed of a large amount of words in the document	Topics composed of few words

Table I. Parameters of LDA and their relevance with respect to documents and topics

With the reference of parameters of LDA as in Table I, LDA model searches for rehashing term designs in the whole Document-Term framework while K-Means grouping relies a lot upon the determination of starting point of convergence [11]. So, in contrast with LDA, K-Means belongs to one cluster for each entity. So, to perform Clustering on the document containing the topics probabilities present in the corpus we are using WEKA. The algorithms can either be applied directly to a dataset or called from your own Java code. WEKA contains devices for information pre-handling, classification, regression, grouping, affiliation rules, and representation. Presently, Applying K-means clustering on output generated from LDA for example, the text record which is showing the breakdown, by rate, of every topic inside every unique text document we imported or we can say the points that form our archives. The next section begins with a review of current studies related to LDA. In Section2, discusses the Literature review, Section 3 discusses the proposed methodology. Section 4 elucidates the experimental setups and Results and Analysis has been presented in Section5. In section 6 we conclude the paper along with suggestions for future research opportunities.

#### II. LITERATURE REVIEW

This section summarizes a few scholarly works proposed about extracting the topics using Latent Dirichlet Allocation (LDA) and various types of clustering used to cluster similar data points. Related work can also be found in the above mentioned applications in multi-domain data. Authors in a research paper [2], proposed an inference algorithm for LDA results that leads to identify documents and the relationships between the documents. [12], applied initial clustering center for K-Means as preprocessing which further makes the latent clustering center more focus on a certain topic.[10], proposed multi-grain clustering topic model which integrates topic modeling and document clustering together. [13], tried to combine LDA with document-specific word distributions for organizing large documents archives. [14], evaluated LDA with the perspective of classifying and identifying the noteworthy topics further applied to filtered collection of Twitter. [1], proposed a generative statistical model, which projects a document into topic space, and each topic contains multiple words. [15], focuses on calculating the temporal weights to reveal the importance of all the topics extracted from the probabilistic approach. [16], introduced an improved K-Means algorithm solving the limitation of traditional K-Means algorithm in terms of dependency on selecting the initial focal points by combining the largest minimum distance algorithm and the traditional K-Means algorithm. Authors in [17] conducted a comprehensive survey on already existing works. Predominantly, they concentrated on event feature learning that has ability on translating social media data into computer friendly numerical form. In [18], Different methods of document clustering and topic modeling on social media content offers a technique of categorizing, interpreting and understanding the large volume of user generated content. Authors have implemented hybrid models to cluster large document sets. Based on contextual keywords, they optimized the clustering similar

#### III. METHODOLOGY

This section depicts the proposed methodology for finding the probabilities of topic distribution after implementing via Single-Link Type, Complete-Link Type and Average-Link Type Hierarchical clustering. The Process diagram for implementation is in Fig. 1.

- A. **Single Linkage Type:** In this approach, we calculate the similarity for the closest pair. The limitation for this method is that close pair groups merge faster than the optimal, regardless of whether there is dissimilarity for those groups.
- B. **Complete Linkage Type:** This approach is based on the phenomenon for calculating similarity of distant pairs along with the limitation of having outliers which causes less merging than optimal one.
- C. Average Linkage, or group linkage Type: In this, grouping of objects are considered for calculating the similarity, rather than individual objects.



Fig. 1. Process Flow Diagram

The Process flow diagram as in Fig.1 starts with the structured data generated after web scraping. The extracted HTML code from web scraping, we stored the data in the database. The generated dataset is then normalized and LDA is applied to extract the topic keywords and probabilistic distribution-based topic composition. The documents obtained after implementing LDA, passed through soft clustering techniques which results in clustering of documents with topic distribution classified against the types of hierarchical clustering as Single-Link type, Complete-Link type and Average- Link type clustering. The experiments have been carried out on WEKA. In this paper, the dataset has been generated by using the BeautifulSoup4 Python tool of web scraping; the python script has been employed to generate the dataset. The generated dataset in the proposed method has

©"BBIJTM July-Dec. 2021", All Rights ReservedPage 21

been stored in two text files that comprises top keywords for each topic in one file and weight distribution of each topic in another file.

#### IV. RESULTS AND ANALYSIS

The output using the commands in MALLET is stored in 2 files. 1st file- keys\_n1.txt is representing what all the top keywords are for each topic as in Fig.2 and the 2nd file- composition\_n1.txt indicates the breakdown, by weight distribution, of every topic extracted from original textual file as in Fig.3.

2	À.	1	C	D	E	F	G	Ħ	I.I.	1.1	K	d (		M	N	0	. 1	Q	8	5	1	U
1	0	0.14285 jea	alousyel	ectronico	ommerce	business	infidelity	exual e-o	mences	es men o	omputers	ervicesv	omen in	ternet	emotiona	products	permissi	on studies	orline jour	nal differe	ences	
2	1	0.14285 ter	chnology	historyn	nemony ci	mputers	oftware co	mputting s	istems res	il-time de	velopme	ntindustr	y time p	rogram	ning ques	tionsper	ormance	engineerin	g science p	roduction	Nanagerre	entken
3	2	0.14285 life	ecellori	ingyreu	sa ma fig	ure mode	s iem ene	gy singula	rity evolu	tion dha n	nocieled n	ature cyc	epartid	ebiole	ectrogyne	natter			1		1,552	
4	3	0.14285 rei	aitty univ	erse synt	axsyntac	tic princip	le languag	e tellic info	rmation lo	igic sespl	state logic	aliniom	ationals	ecursion	n medium	states la	s duality	rules chru				
5	4	0.14285 dis	sease der	nentia al	zheimerå	C"s doi e	mergingdi	nical cogn	itive disea	ses mci in	fections f	lactors hu	manhea	Ith impa	iment d	agnosis b	rain erne	rgence pati	ents infect	icus mild		
6	5	0.14235 ros	ute netw	orkpacks	thost res	earch nel	workshos	s request	data comr	nunication	n spectrum	n touting	radio pro	ctocol to	chnologi	es packet	designs	ervices cac	he discore	γ		
7	5	0.14285 pa	rty partile	spolitica	lvoters	roups pol	icy politics	candidate	samericar	n electora	activists	press can	didate di	emocrat	liccoalitie	n theory	election	coliticiens p	iositi ons n	omi nation	5	
8	7	0.14205 se	curity stu	cies war	internati	onal milit	ary politica	national	ficiatorsh	ips politic	sdemocra	cypolicy	regimes	norld a	old field :	cholars e	conomic l	foreign dor	nestic resea	arch	13	
9	8	0.14285 fei	ature fea	tures dat	adusteri	ng duster	s selection	criterion s	ubset nur	rbendass	dusterer	rcr fixed	method	subsets	unsuperv	ised fsse	n results	figure met	nods			
10	9	0.14285 th	eory form	structur	e science	nature sy	sten gene	ral model	potential i	informatio	nevoluti	or existe	nce theo	iles rela	tionship	explanat	ion conce	pt fundam	ental natur	al adst in	temai	
11	10	0.14285 pc	liceright	s hungan	rtia euro	pean publ	icratism n	ima repor	racist dis	mininatio	a BLEECE	member	numan n	ational	orime stat	esintole	ance hate	eunion				
12	n	0.14285 mi	andelalin	ing activ	ty univer	sity accel	eration rec	grition g	svity trial	bodyane	motion af	maend	ineyea	rs gyrosi	ope acis	dataset ch	aracteris	tics south			1	
13	12	0.14285 pr	eycarriv	prous pla	nts plant	species d	ata sarrace	n'a utricula	iria traps c	irosera el	lison darw	in captur	e camivo	xygene	ra pingui	ula purp	realanah	sis nepent	hes got elli			
14	13	0.14285 lea	adership	ethical in	ternation	al leader	legalizatio	n global n	пападетте	nt legal d	evelopme	int leader	busines	s ethics	values or	ganization	organiza	tional jour	nal effectiv	e issue en	nployees w	lume
15	14	0.14285 cas	se set co	ות חכותו	ike made	point fem	n single re	sult standa	ind fact con	mplex pro	blem bad	k state ad	ditional	hand po	ints limit	ed effect		1				
16	15	0.14285 lea	arning ma	xhine da	ta trainin	g algor ith	ns decision	ezample	s algorithm	n dassific	ation prot	len met	nods set	neurals	lassifiero	lass featu	res netw	curks classif	ers trees in	istances	1	
17	16	0.14285 izv	vorinina	l fundam	ental just	ice office	related ca	es govern	ment enfo	incernent t	court supp	port coop	eration s	special b	reatment	racial per	sons rega	rd article n	eleventleg	islation		
18	17	0.14285 the	ecrygeo	netry cer	ntury mat	hematics,	proups too	ballthing	s physics +	ork algeb	ra linear :	space lie	alled str	cry quar	tum func	tionsdim	ensions e	quations p	arts			
19	18	0.14235 tra	its varial	ion leaf g	eneticse	lection sp	ecies plan	evolution	plants ad	aptive fit	ness ecop	rysiologi	al enviro	onment	s evolutio	пату рор	dations p	lasticity na	tural patter	ms respon	ses studies	
20	19	0.14285 sp	edficart	de defin	ite tems	sentence	reacing er	glish cond	ttice time	group na	tve learn	ers conte	d article	s indefi	n te advar	naed kore	an specifi	atty speaks	rintented	ete		
21	20	0.14285 mi	ecia tele	ision eff	ects viole	nce softw	are source	shareope	ratingbro	NSEIS SYS	tems mas	stheory	nerket gr	owth ca	en sacie	y social v	idient aga	ressive mo	bile			
22	21	0.14285 an	imals hu	nen mor	ai rights h	umansan	imal resea	thsenses	ubjects be	ingsyord	mind con	sequero	s thoma	s wrong	species	we books	lives long	5				
23	22	0.14285 do	main dre	domain	reputati	onretwo	rk names fi	satures ma	licicus no	tos legitin	nate ris se	tfguren	umberv	ectors i	os sectilor	informat	ion zone	database		l		
e1	- 11	A 41937 ala				41	h								وترجعناهم				aking.			
- (4	Ъ.	keys n1	Ŧ											•							_	•
READ!																		ŧ	3	<u> </u>	1	+ 10

Fig. 2. Top Keywords

		0	D	Е	ा	G	н	. в.	E	ĸ	τ.	M	N	C	LE
1 docho	docName	t1	12	13	ta	15	tš	17	18	13	110	111	112	113	u
2	0 file/C/malet/sample-data/web/data/08_LRGL5_A04_68.txt	0.458399	0.014313	0.003324	9.21E-05	7.39E-D4	0.019485	9.21E-05	0 017545	0.001385	0.011061	9.235-05	9 238-05	9.23E-05	5 0
3	1 file /C/malet/sample-data/web/data/10.1.1.307.5273.bt	8.41E-05	0.027347	8 415-05	8.41E-05	8.41E-05	0.024605	8.41E-05	8.415-05	6.735-04	8.41E-05	0.004204	8.415-05	8.41E-05	5 0
4	2 file/C/mailet/sample-data/web/data/11-36-1-PB.txt	3.78-00	3.78E-05	3 785-05	0.001516	3.388-05	3.025-04	3.78E-00	3.785-05	3.788-03	0.032586	0.002155	3.788-05	3.78E-0	5 0
5	3 file /C /mailet/sample-data/web/data/1115-d165-1-P8 trt	0.465340	9.458-05	0.00063	9.45E-05	0.095133	4.00737	0.002079	9.456-05	9.456-05	7.568-04	9.456-05	0.001417	9.45E-05	5 4
6	4 file /C/mallet/sample-data/web/data/1251-3556-1-PB.tat	0.013473	0.033499	8076-05	B.DTE-05	0.003471	8.076-05	8.0TE-05	0.004135	6.466-04	0.0159	8.075-05	8.076-05	0.402307	5 8
7	5 file/C/mailet/sample-data/web/data/148-208-1-PILtxt	1.848-05	6.62E-04	1.845-05	1.848-05	1.84E-05	2.756-04	7.916-04	5.338-04	0.019971	0.422575	1478-04	1848-05	1.4E-0	5 1
9	6 file/Cy/mailet/sample-data/web/data/22.txt	0.003018	8.38E-05	8 385-05	0.002431	8.38E-05	8.385-05	8.3IE-05	0.023427	8.382-05	0.026492	8.385-05	8365-05	8.38E-05	5 E
9	7 file/C/mallet/sample-data/web/data/29971-29673-1-98.txt	8.31E-05	0.008104	0.002409	8,316-05	0.017525	8.316-05	0.005116	0.002409	8.316-05	8.J1E-05	8.310-05	8316-05	8.J1E-05	5 E
C	8 ille/C/mailet/sample-data/web/data/3.txt	0.914173	0.045202	8 555-03	8.31E-00	0.023338	0.022045	8.31E-03	8.535-03	8.9%-03	0.013734	0.001909	8.956-03	0.00322	2 8
1	s file/c/mallet/sample-data/web/data/seu3148.txt	4.178-00	417E-00	41/6-05	\$.1 TE-10	0.004482	0.010851	4.1TE-10	4.176-00	4.1/2-03	0.010094	0.004424	4178-03	0.0038	4 5
2	10 file /C //mailet/sample_data/web/data/9780472068958-3 tot	1.75E-04	0.015487	0.001399	1.75E-04	0.001399	1.755-04	1.75E.04	0.014163	1.756-04	0.467494	1.756-04	0.046585	1.75E-0	4 0
3	11 file/C/mailet/sample-data/web/data/Ackerly_etal_2003.bit	2.146-05	2.14E-05	0.001372	1.715-04	2.34E-05	2.146-05	3.22E-04	2.146-05	7.725-04	0.419226	3.225-04	2.145-05	0.41232	5 5
4	12 file/C/mailet/sample-data/web/data/A).txt	5.21E-04	0.000159	6 215-04	5.21E-04	0.009317	0.013665	6.21E-04	0.21E-04	6.218-04	0.457143	6,215-04	6 215-04	6.21E-0	1
5	13 file /C /mailet/sample-data/web/data/MRosNegFactor.tet	0.043515	0.009121	1.515-05	0.002113	9.665-01	0.000662	0.001706	0.003192	0.000074	0.036758	1.516-05	1 315-04	1.51E-05	5 0
6	14 file/C/mallet/sample-data/web/data/An Introduction Te-Electr	0.414019	0.045366	1075-04	0.001504	1.07E-01	0.012562	1.07E-04	0 017322	B 556-04	0.426305	1.075-04	1.075-04	1.176.07	1 0
7	15 file /C/mallet/sample-data/web/data/Antonatakis.txt	2.25E-05	2.25E-05	2 256-05	2.256-05	2.25E-05	0.004122	2.25E-05	1.395-04	0.011531	0.405468	2.255-05	2 256-05	2.25E-05	5 2
8	16 file /C /mailet/sample-data/web/data/anyah2ithcentury.txt	4.118-05	3.35E-04	6 285-04	4.118-05	3.15E-04	4.155-05	4.188-05	4.385-05	4188-05	0.040464	4.385-05	6 28E-04	4.18E-05	5 3
¢	17 file /C /mallet/sample-data/web/data/automat 2018 17.1 29.trt	5.71E-05	0 574116	5746-05	5.7/E-05	5.ME-05	0.000469	5.74E-05	4.596-01	0.060469	0.011311	5.746-05	5716-05	0.001263	1 5
0	18 file/C/mailet/sample-data/web/data/856-Vision for Researchy	0.011558	0.015739	1.306-05	2.866-04	1.XE-05	0.460425	9.21E-04	1.040 04	7.406-04	0.437373	1.305-05	1.046 04	1.HE 0	1 0
1	19 file/Cy/mailet/sample-data/web/data/bactena.txt	4.73E-04	4.75E-04	0.040555	4.73E-04	0.09182	4.758-04	4.73E-04	0.003797	4.758-04	4.73E-04	4.358-04	0.007115	0.0015	24
2	20 file /C /mailet/sample-data/web/data/Baldwin%20(1995)%20Sec	2.525-05	0.001158	2.625-03	2.53E-05	0.002403	2.625-05	2.5JE-05	0.683973	2.625-05	0.026409	7.595-04	2.62E-05	2.09E-0/	1 0
3	21 file/C/inalit/sample-data/web/data/Balogh_2000_ApJ_540_113	4.0LE-05	4.01E-05	4016-03	4,010-05	0.003405	4.018-05	0.001723	0.002844	0.002283	0.421628	4.010-05	4.016-03	3.20E-04	14
1.1	composition_n1 ①						1						1.000 00		
wor											E E		- 1		133

Fig. 3. Topic Composition

Preprocess Classify Cluster Associate Select	tahibukes Visualize
lusterer	
Choose SimpleKMeans init 1 -max candidates 1	00 - periodic-pruning 1 0000 - min-density 2.0 -t1 -1.25 -t2 -1.0 -N 10 -A "webs.core EuclideanDistance -R first-last" -I 500 - num-slots 1 - 9 11
luster mode	Obsterer output
Use training set	Tau
O Supplied test set Set.	151
O Porcentage apit % 66	133 104
O Classes to clusters evaluation	638
(Num)135 *	
Store clusters for visualization	
Ignore attributes	Time taken to build model (full training data) : 0.05 seconds
Start Stop	Model and evaluation on training set
esuit list (right_click for options)	Clustered Instances
14 23 05 - SimpleKNeans	0 3 ( 51)
1424.12 - SimpleKWeans	1 22 ( 348)
14.25:15 - HierarchicalClusterer	3 2 ( 31)
14.26.24 - HierarchicalClusterer	4 1 ( 23)
14.26.35 - HistoricalClusterer	5 2 ( 34)
14.26.07 - dir planaround	
	E 4 ( 43)
	9 20 ( 311)





Fig. 5. Cluster of Documents with topic distribution present in the document corpus

We can also visualize the clusters that are formed. The clusters formed after applying K- means clustering on the topics composing the documents as in Fig. 5.

( Cluster (Norrý	Y. todvo (Hum)	
lakut Oluster Alumi 🔹	Selectionarce	1
mosat Dear Open Save	Jter	
st:composien_nt_clastend	12.5	
n I	9 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8	
elument elument elument	ervents coverts tracely	- State
ass colour		

Fig. 6. Single-link type Hierarchical clustering

The results of the Single link, complete link and Average-link of the Hierarchical clustering are as in Fig 6, Fig.7 and Fig. 8.



Fig. 7. Complete-link type Hierarchical clustering

Custer (Nom)		1	Y docho (Num)			
lolour: Clusler (Nom)			SelectInstance			
Rest	Cikar Open	Sav			Jiller	
t composion_n1_dustated						
3 * * * * *					4 15	Y
	•					
	1					- 9410 (16) - 7 - 7 - 7 - 7 24 4
cluster) cluster)	choise) choise)	alarters.	ciuter durant	e (senade	Costur(s	No. 1 Ang ang ang ang ang ang ang ang ang ang a

Fig. 8. Average-link type Hierarchical clustering

#### V. CONCLUSION

From the experiments, it has been concluded that the clusters are made on the basis of the distribution of topics over which the documents are close to each other in terms of their meanings and the probability value is same for Complete-Link Type and Average-Link Type clustering for each topic in each document. Less-fold cross validation might be the reason for the same values in the probability values. In contrast to this, Single- Link type has very low probability for topics. As Topic modeling is an emerging and current field, we can increase the cross validation on dataset to have more accuracy and we can also approach different methods to improve the performance of the models for managing large archives of information. The futuristic directions for further research can Fuzzy logics and the Dynamic data-set. We can perform a hybrid approach by combining LDA and K-Means clustering on a dynamic dataset instead of static. In the Fuzzy Logic approach the simple LDA topics can be considered in a crisp set. Hence, we can fuzzify the output in results; and further the recovered topics can be in the form of a fuzzy set.

#### REFERENCES

- [1] D. M. Blei, A. Y. Ng, and M. I. Jordan, "Latent Dirichlet allocation," J. Mach. Learn. Res., vol. 3, no. 4–5, pp. 993–1022, 2003, doi: 10.1016/b978-0-12-411519-4.00006-9.
- [2] M. Griffiths, T.L.; Steyvers, "Finding scientific topics," in Proceedings of the National Academy of Sciences, 2004, pp. 5228--35.
- [3] D. M. Blei, "Probabilistic Topic Models," pp. 77-84.
- [4] D. Blei, L. Carin, and D. Dunson, "and applications to document and image analysis ] [ A focus on graphical model design," no. November, pp. 55–65, 2010.
- [5] J. Chien, S. Member, and C. Chueh, "Topic-Based Hierarchical Segmentation," vol. 20, no. 1, pp. 55–66, 2012.
- [6] J. Chien, S. Member, C. Chueh, and S. Member, "Dirichlet Class Language Models for Speech Recognition," vol. 19, no. 3, pp. 482–495, 2011.
- [7] J. Chien and M. Wu, "Adaptive Bayesian Latent Semantic Analysis," vol. 16, no. 1, pp. 198–207, 2008.
- [8] Y. Chang and J. Chien, "LATENT DIRICHLET LEARNING FOR DOCUMENT SUMMARIZATION," pp. 1689–1692, 2009.
- [9] N. Chiao, "HIERARCHICAL THEME AND TOPIC MODEL FOR SUMMARIZATION Jen-Tzung Chien and Ying-Lan Chang Department of Electrical and Computer Engineering," 2013.
- [10] E. P. Xie, P.; Xing, "Integrating Document Clustering and Topic Modeling," 2013.
- [11] W. H. Li Y., "A clustering Method Based on K- Means Algorithm," in 2012 International Conference on Solid State Devices and Materials, Science Direct: Physics Procedia 25, pp. 1104–1109.
- [12] Z. Guan, P.; Wang, M.; Chen, B.; Fu, "K-means Document Clustering Based on Latent Dirichlet Allocation," Forty-Fifth Annu. Meet. West. Decis. Sci. Inst., 2016.
- [13] M. Chemudugunta C.; Smyth P.; Steyvers, "Modeling General and Specific Aspects of Documents with a Probabilistic Topic Model," Adv. neural Inf. Process. Syst., 2006.
- [14] O. D.A., "Using Latent Dirichlet Allocation for Topic Modeling in Twitter," in In: Proceedings of the 2015 IEEE 9th International Conference on Semantic Computing, CA, USA, 2015, pp. 978-1-4799-7935–6.
- [15] L. J. Z. D. Chen H., Zhang G., "A Fuzzy Approach for measuring development of topics in patents using Latent Dirichlet Allocation," in In: 2015 IEEE International Conference on Fuzzy Systems, 2015, p. ISBN: 978-1-4673-7428-6., doi: 10.1007/s10489-014-0595-0.
- [16] Z. Yang, Q. Li, Z. Lu, Y. Ma, Z. Gong, and H. Pan, "Semi-supervised Multimodal Clustering Algorithm Integrating Label Signals for Social Event Detection," Proc. - 2015 IEEE Int. Conf. Multimed. Big Data, BigMM 2015, pp. 32–39, 2015, doi: 10.1109/BigMM.2015.26.
- [17] H. Zhou, H. Yin, H. Zheng, and Y. Li, "A survey on multi-modal social event detection," Knowledge-Based Syst., vol. 195, p. 105695, 2020, doi: 10.1016/j.knosys.2020.105695.
- [18] S. A. Curiskis, B. Drake, T. R. Osborn, and P. J. Kennedy, "An evaluation of document clustering and topic modelling in two online social networks: Twitter and Reddit," Inf. Process. Manag., vol. 57, no. 2, 2020, doi: 10.1016/j.ipm.2019.04.002.
- [19] S. A. Devi and S. S. Kumar, "A hybrid document features extraction with clustering based classification framework on large document sets," Int. J. Adv. Comput. Sci. Appl., vol. 11, no. 7, pp. 364–374, 2020, doi: 10.14569/IJACSA.2020.0110748.



"Bodh", BPIT's International Journal of Technology & Management ISSN: 2454-8421, Volume 7, Issue 2, July- Dec 2021, Page 26-30

## A Way to Trust Deportment for Security in Distributed Systems Preeti Arora<sup>1</sup>, Joy Chugh<sup>2</sup>, Preeti<sup>3</sup>, Tarun Tomar<sup>4</sup>

<sup>1,2,3</sup> Department of Computer Science and Engineering, Bhagwan Parshuram Institute of Technology, Delhi, India.

<sup>1</sup>erpreetiarora07@gmail.com, <sup>2</sup>joychugh33@gmail.com, <sup>3</sup>preetihinwal2409@gmail.com, <sup>4</sup>taruntomar022@icloud.com

Abstract -The objective of this paper is to describe the implementation of the 3D model of a library using Virtual Reality. VR is emerging as an interesting new technology in today's world where we can experience a 3D environment from anywhere. VR experiences can build inordinately immerse environments and spots, asserting users to move in that place without actually visiting there. In this project, a 3D model of our college library full of interactive objects [3] is made in which the user can move around and explore the environment. It includes various scripts written in C# programming language to help it become interactive. Gaming engine Unity was used to develop this project.

Keywords— Virtual Reality, Unity, Script, Headset

#### I. INTRODUCTION

Virtual Reality (VR) provides a simulated experience that is almost identical to the real world. VR provides interactive content which enables the user to explore the entire 360 degrees of a scene. From conventional classroom place to utmost training spots, VR down scales investment and enriches advancement through a range of industries. VR requires a headset to get the immersive experience to see the 3D environment. During this pandemic situation when people are not able to go out and experience the place, they can enjoy the 3-D model of the place which is almost identical to the real world and can utilize this model for their work. Nowadays Libraries [11] have become a kind of a community center centered on knowledge and education [8]. Providing an almost identical experience to the real world with Virtual Reality sounds like science fiction but isn't it cool, people can move freely and experience the place virtually [1].

Virtual Reality technologies can be applied to college libraries to give the users a 3-Dimensional experience of the library where the user can move around and explore the library virtually. The user can interact with the 3D objects present in the library virtually, thus this application can act as a tool to maximize the operability of the library, where interactive objects can enrich the user's experience and inspire users to visit the library more often. This model helps the users to get the information and navigate easily using VR based mobile applications when moving inside the library. The 3D model of the library makes it easier for the user to explore the library without visiting in person. This project uses scripts written in C# language to make the application [6]. The findings of our study provide a contribution to the research field concerning user's perceptions of benefits of smartphone-based virtual reality applications within the education industry [12] where the user can get an immersive experience. Using this 3D model, the user can see where the different book-sections are located, and the smartphone-based virtual reality application will help the user to move in the different sections inside the place. VR has signified its ability to transFig and manage results in the learning and instruction arena. There are different types of headsets [10] available in the market today like oculus rift, Samsung gear etc.

#### II. LITERATURE REVIEW

In past research the visualization of complicated engineering data has been done and a set of multimodal tools has been provided to audit a 3D model on various levels, discussing the difficulties that arose while the implementation process and the future applications of VR in the industrial sector. We have implemented one of those applications in the education sector. Virtual Reality can act as an amazing medium to connect people to the unreachable world where people can experience the real world using the virtual environment created using VR. This 3-D world not only provides the immersive experience to the real-world but also the user can interact with the objects present in the 3-D environment. Virtual World has reached the healthcare sector [5] and now it is the turn for the education sector. Education sector needs the implementation of 3D models, helping students in increasing their knowledge and improving the future. We have presented the development of Augmented Reality and Virtual Reality to help users to navigate [2] and get the information easily using VR based mobile applications while walking inside the library. This paper introduced VR technology to enhance and improve the user experience inside the library in the right direction and with the right information instantly [3] addressed the topic to give the third degree to the notion of virtual reality and its suggested uses in libraries to shore the teaching of fact literacy.

#### III. PROPOSED WORK

Fig 1 depicts the flow of the proposed model, starting from how we created the 3D environment to how we added objects and interactions in the model. Data collection [6] is the first step for modeling any 3D environment. After data collection the design is built and the objects are created and added in the 3D environment. For adding interactions to the objects a script in C# has to be written. Fig. 1 depicts the library layout.



Fig 1: Layout of Library



Fig 2: Flow chart for Creation of 3d model of Library

The first step to do in the model is to implement the layout of the library.



Fig 3: Texture of wall

A tour of the library was done and the data was collected [4] for different materials and texture of library objects. Material is the basic look which tells what the object is made of. Texture on the other hand provides it a more realistic feel. Texture mapping [9] was done to make this happen as in Fig.3.

A C# script is attached to the main camera to control the moving character. Logic for script (if camera angle is greater than 25 degree):

- Toggle angle = 25 degree
- Moving speed = 3.0f
- Vector3forward = vrCamera.TransformDirection(Vector3.forward)
- move = forward \* speed



Fig 4: Background Lighting

Character controller for the main camera acts like a 3D character mesh which is attached to the main camera game object. It acts like a rigid body without the need for colliders is as shown in Fig 5.



Fig 5: Character Controller for main camera

### IV. RESULT AND DISCUSSION

Virtual reality provides an environment [7] which is presented to the user in a way that their precept is deferred and they accept it as the real environment. On a device, virtual reality is primitively experienced through the two of our five senses that is "sight" and "sound". With this project we have created a 3D virtual environment for a library in which the user can freely move around and interact with different interactive objects. It provides the user a fresh and new way of being in a place just while sitting at home or anywhere by putting on the headset and enjoying the environment around them.

#### V. CONCLUSION AND FUTURE WORK

Whilst the VR concept isn't new, the technology to power the VR is, as this technology delivers a seamless experience for the user. The bright side is that with nearly limitless applications, VR offers an abundance of opportunities in almost every aspect of our world, including education, design, commerce, gaming, tourism, medicines [5] and knowledge. Since many of us have already created things that never leave the web's digital space, it's not facile to imagine a primarily digital future which will be powered by a VR environment. Virtual Reality is no longer a fiction, it is integrated into our present and, in the future it will lead to a lot of advancement that will shape the future and it will be very beneficial to the industry. Some interactions can be added to the model in future which makes the user able to read books from the shelves.

#### REFERENCES

- [1] Wolfartsberger, Josef & Zenisek, Jan & Sievi, Christoph & Silmbroth, Mathias. (2018). A Virtual Reality Supported 3D Environment for Engineering Design Review.
- [2] Romli, Nida & Razali, Amir & Ghazali, Nur Hafizah & Binti Zahri, Nik Adilah Hanin & Ibrahim, Siti. (2020). Mobile Augmented Reality (AR) Marker-based for Indoor Library Navigation. IOP Conference Series: Materials Science and Engineering.
- [3] Massis, Bruce. (2015). Using virtual and augmented reality in the library. New Library World.
- [4] Lee, H-L. (2000). What is a collection? Journal of the American Society for Information Science, 51(12), 1106–1113.
- [5] Carson, E. (2015). 10 ways virtual reality is revolutionizing medicine and healthcare. TechRepublic. Retrieved from http://www.techrepublic.com/article/10-ways-virtual-reality-is-revolutionizing-medicine-andhealthcare/
- [6] Wilson, C. J., & Soranzo, A. (2015). The use of virtual reality in psychology: A case study in visual perception. Computational & Mathematical Methods in Medicine, 2015, 1–7.
- [7] Mazuryk, Tomasz & Gervautz, Michael. (1999). Virtual Reality History, Applications, Technology and Future.
- [8] Hahn, J. (2015). The student/library computer science collaborative. portal: Libraries and the Academy, 1(2), 287–298.
- [9] Donalek, C., Djorgovski, S. G., Cioc, A., Wang, A., Zhang, J., Lawler, E., Yeh, S., Mahabal, A., Graham, M., Drake, A., Davidoff, S., Norris, J., & Longo, G. (2014). Immersive and collaborative data visualization using virtual reality platforms. In Big Data (Big Data), 2014 IEEE International Conference on (609–614).
- [10] Reisinger, D. (2016). Should you buy an Oculus Rift VR headset: 10 Factors to consider Eweek, March 30, 7.
- [11]. Mikkelsen, S., & Davidson, S. (2011). Inside the iPod, outside the classroom. Reference Services Review, 39(1) 66–80.
- [12] Aji, C. A., & Khan, M. J. (2015). Virtual to reality: Teaching mathematics and aerospace concepts to undergraduates using unmanned aerial systems and flight simulation software. Journal of College Teaching & Learning, 12(3), 177–188.
- [13] Tharp, Gregory, Virtual Reality: A New Dimension for Academic Libraries (2019).
- [14] "Special study on virtual reality technology: virtual reality head-mounted technology, display and interactive device" by sraSontisirkit. Asian institute of school of engineering and technology, Thailand.
- [15] "Virtual Reality in Libraries" by Breanne Kirsch (Briar Cliff University, USA), 1-33 (information available at https://www.igi-global.com/chapter/virtual-reality-in-libraries/255391).
- [16] "Mobile Virtual Reality featuring a six degrees of freedom interaction paradigm in a virtual museum application" by M. Papaefthymiou, K. PlelisD. Mavromatis, G. Papagiannakis
- [17] Student Engagement and Smart Spaces: Library Browsing and Internet of Things Technology by Jim Hahn (University of Pennsylvania, USA)



"Bodh", BPIT's International Journal of Technology & Management ISSN: 2454-8421, Volume 7, Issue 2, July- Dec 2021, Page 31-44

## Light Stimulated Rat (LSR) Algorithm Based Multimodal Image Fusion for MRI Images Jayant Bhardwaj<sup>1</sup>, Pawan Sharma<sup>2</sup>

<sup>1,2</sup>Department of Electronics and Communication, Bhagwan Parshuram Institute of Technology, Delhi, India, <sup>1</sup>jayantbhardwaj@bpitindia.com, <sup>2</sup>pawan061971@gmail.com

Abstract - Fusion of MRI images enhances the level of information in a great way. The features of individual images not only get clubbed but the diagnostic approach for diseases will also get increased. In this research a detailed study on the image fusion strategies in medical images has been carried out. In this research paper a novel bio inspired optimization Light stimulated rat algorithm has been employed for Multimodal MRI Images Image Fusion. The proposed work has been compared with the state-of-the-art Fusion method and it has been identified that the proposed work out performs with great merit over the other fusion methods and the proposed research work can be considered novel for the transform based medical image fusion schemes. Discrete wavelet transform based MRI decomposition and reconstruction scheme has been employed in this proposed work.

Keywords--- fusion, information, medical images

I. INTRODUCTION

In the direction of diagnostic analysis with different medical modalities sufficient efforts are being done to extract meaningful features and hence information in the form of digital images so that any decision can be established about an ailment in the body[1],[2].From many years a lot of image processing techniques like image deblurring, image enhancement, denoising etc. have been evolved to extract out maximum information from the source medical images[3],[4].Multimodal Image fusion is also an important and growing technique in this domain. Two modality images are so fused with each other pixel wise that the third image consists of the features of both images [5], [6], [28],[29].The motive of this pixel level image fusion method is to eliminate the redundancy of image feature and common information and to reserve the complementary information from both images to the third image i.e. fused image[7],[8]. This process of image fusion not only conserves the memory of a computer system but also helps in reduction of decision time of a doctor or of an automatic disease diagnosis system[9],[10],[26]. Every modality of an image technology such as MRI, generates an image with a salient feature or information [11],[12],[27].

Some modalities provide information about the peripheral region of organs while some provide detailed information about central features of an organ. In case of human brain MRI imaging Flair and T2 are such two modalities for these purposes respectively.T1C and T1 are also two important modalities which provide the fluid information and information regarding minute discontinuities in the tissues of the brain respectively. In transform domain fusion methods such as wavelet based, contourlet(CT) ,non-subsampled contoulet (NSCT) and curvelet transform (CVT) etc. the signal processing like image decomposition at frequency level is performed on image coefficients .According to information measure ,the fusion technique is broadly grouped in three ,that is pixel, feature and decision. In the pixel class the direct operations are performed

on pixels of images to fuse[13],[14].In the feature class the operations are performed at frequency level to enhance the detailed prime features of images like edges, points and corners etc. At the decision class the images are fused by applying classifier rules and fuzzy logic etc. to the image coefficients [15],[16]. Fig.1 and Fig.2 provide a conceptual view about the medical modalities of MRI.



Fig.2.Dataset-2 modalities

The fusion result for above Fig 2 is shown in the result section in fig.7.

#### II. METHODOLOGY

This section discusses the methodology employed using Image Decomposition and Fusion rule

#### A. Image Decomposition

The methodology used here is basically a signal decomposition into different coefficients of frequency for individual images and further mix them into each other with in limits of signal processing and its operators[17],[18]. Fig.3 presenting the Schematic diagram of the proposed Image Fusion method.



Fig.3 Schematic diagram of the proposed Image Fusion

#### B. Fusion Rule

In the fusion rule a metaheuristic scheme for optimization is applied. The strange behaviour of rats with exposure to light is mimicked in the fusion model [19-21]. The MI of each pixel is calculated and the way a rat finds the maximum displacement on light stimulation is modeled to raise the value of MI per pixel. Further the pixels are clubbed to each other and the final image reconstructed after the process of reconstruction by some appropriate transform tool [22-25].

#### **RESULT AND DISCUSSION**

The results for the proposed fusion scheme are illustrated in Fig. 4 to Fig. 7. The results have been evaluated qualitative as well as quantitatively. The evaluation measures values of mutual information and other parameters like RMSE and PSNR proves the merit of this fusion method. Two data sets from a famous MRI data set are tested .The proposed method is compared with three very renown techniques of fusion. These are Wavelet and Halo-whale method, SWT-NSCT combined method and simple NSCT method. Our proposed method is outperforming the compared one method. The effectiveness of the proposed method is deliberately explained in this section. Table I and Table II are providing a view of satisfactory results.

Metrics	Wavelet + HW Fusion	SWT + NSCT	NSCT	Proposed
Mutual information	1.4673	1 4669	1.4299	1.5765
PSNR (in ffi)	37.8289	37.8848	36.8904	44.0957
RMSE	6.5 <mark>1</mark> 01	6.5281	6.5409	5.4940

 Table 1 Comparative analysis using dataset-1

 Table 2 Comparative analysis using dataset-2

Metrics	Wavelet + HW Fusion	SWT + NSCT	NSCT	Proposed
Mutual information	1.4612	1.4550	1.4508	1.4960
PSNR (in dB)	32.991	33.470	32.498	35.541
RMSE	10.0053	10.0510	9.9689	<b>9.640</b> 4


©"BBIJTM 2022", All Rights ReservedPage 37



©"BBIJTM 2022", All Rights ReservedPage 38



©"BBIJTM 2022", All Rights ReservedPage 39



Fig.7 MRI Fusion

## REFERENCES

[1]S. Singh and D. Gupta, "Detail Enhanced Feature-Level Medical Image Fusion in Decorrelating Decomposition Domain," IEEE Trans. Instrum. Meas., vol. 70, no. November, 2021, doi: 10.1109/TIM.2020.3038603.

[2] A. V. Nikolaev et al., "Quantitative Evaluation of an Automated Cone-Based Breast Ultrasound Scanner for MRI-3D US Image Fusion," IEEE Trans. Med. Imaging, vol. 40, no. 4, pp. 1229–1239, 2021, doi: 10.1109/TMI.2021.3050525.

[3] K. He, J. Gong, L. Xie, X. Zhang, and D. Xu, "Regions preserving edge enhancement for multisensor-based medical image fusion," IEEE Trans. Instrum. Meas., vol. 70, 2021, doi: 10.1109/TIM.2021.3066467.

[4]N. Geng, Z. Chen, Q. A. Nguyen, and D. Gong, "Particle swarm optimization algorithm for the optimization of rescue task allocation with uncertain time constraints," Complex Intell. Syst., vol. 7, no. 2, pp. 873–890, 2021, doi: 10.1007/s40747-020-00252-2.

[5]K. Mishra and S. K. Majhi, "A binary Bird Swarm Optimization based load balancing algorithm for cloud computing environment," Open Comput. Sci., vol. 11, no. 1, pp. 146–160, 2021, doi: 10.1515/comp-2020-0215.

[6]O. S. Faragallah et al., "A comprehensive survey analysis for present solutions of medical image fusion and future directions," IEEE Access, vol. 9, no. Januar. 11358–11371, 2021, doi: 10.1109/ACCESS.2020.3048315.

[7] M. Arif and G. Wang, "Fast curvelet transform through genetic algorithm for multimodal medical image fusion," Soft Comput., vol. 24, no. 3, pp. 1815–1836, 2020, doi: 10.1007/s00500-019-04011-5.

[8] K. Padmavathi, C. S. Asha, and V. K. Maya, "A novel medical image fusion by combining TV-L1 decomposed textures based on adaptive weighting scheme," Eng. Sci. Technol. an Int. J., vol. 23, no. 1, pp. 225–239, 2020, doi: 10.1016/j.jestch.2019.03.008.

[9] D. Xu, Y. Wang, S. Xu, K. Zhu, N. Zhang, and X. Zhang, "Infrared and visible image fusion with a generative adversarial network and a residual network," Appl. Sci., vol. 10, no. 2, 2020, doi: 10.3390/app10020554.

[10] J. Bhardwaj and A. Nayak, "Haar wavelet transform–based optimal Bayesian method for medical image fusion," Med. Biol. Eng. Comput., vol. 58, no. 10, pp. 2397–2411, 2020, doi: 10.1007/s11517-020-02209-6.

[11] R. Hou, D. Zhou, R. Nie, D. Liu, and X. Ruan, "Brain CT and MRI medical image fusion using convolutional neural networks and a dual-channel spiking cortical model," Med. Biol. Eng. Comput., vol. 57, no. 4, pp. 887–900, 2019, doi: 10.1007/s11517-018-1935-8.

[12] I. Aljarah, H. Faris, S. Mirjalili, N. Al-Madi, A. Sheta, and M. Mafarja, "Evolving neural networks using bird swarm algorithm for data classification and regression applications," Cluster Comput., vol. 22, no. 4, pp. 1317–1345, 2019, doi: 10.1007/s10586-019-02913-5.

[13] Y. Wang, Z. Wan, and Z. Peng, "A Novel Improved Bird Swarm Algorithm for Solving Bound Constrained Optimization Problems," Wuhan Univ. J. Nat. Sci., vol. 24, no. 4, pp. 349–359, 2019, doi: 10.1007/s11859-019-1406-6.

[14] Y. Yang, J. Wu, S. Huang, Y. Fang, P. Lin, and Y. Que, "Multimodal Medical Image Fusion Based on Fuzzy Discrimination with Structural Patch Decomposition," IEEE J. Biomed. Heal. Informatics, vol. 23, no. 4, pp. 1647–1660, 2019, doi: 10.1109/JBHI.2018.2869096.

[15] S. D. Ramlal, J. Sachdeva, C. K. Ahuja, and N. Khandelwal, "An improved multimodal medical image fusion scheme based on hybrid combination of nonsubsampled contourlet transform and stationary wavelet transform," Int. J. Imaging Syst. Technol., vol. 29, no. 2, pp. 146–160, 2019, doi: 10.1002/ima.22310.

[16] A. Dogra, B. Goyal, and S. Agrawal, "Medical image fusion: A brief introduction," Biomed. Pharmacol. J., vol. 11, no. 3, pp. 1209–1214, 2018, doi: 10.13005/bpj/1482.

[17] P. H. Venkatrao and S. S. Damodar, "HWFusion: Holoentropy and SP-Whale optimisation-based fusion model for magnetic resonance imaging multimodal image fusion," IET Image Process., vol. 12, no. 4, pp. 572–581, 2018, doi: 10.1049/iet-ipr.2017.0573.

[18] B. J and N. A, "Cascaded Lifting Wavelet and Contourlet Framework Based Dual Stage Fusion Scheme for Multimodal Medical Images," J. Electr. Electron. Syst., vol. 07, no. 04, 2018, doi: 10.4172/2332-0796.1000292.

[19] P. Hill, M. E. Al-Mualla, and D. Bull, "Perceptual Image Fusion Using Wavelets," IEEE Trans. Image Process., vol. 26, no. 3, pp. 1076–1088, 2017, doi: 10.1109/TIP.2016.2633863.

[20] A. Dogra, B. Goyal, and S. Agrawal, "From multi-scale decomposition to non-multi-scale decomposition methods: A comprehensive survey of image fusion techniques and its applications," IEEE Access, vol. 5, pp. 16040–16067, 2017, doi: 10.1109/ACCESS.2017.2735865.

[21] F. E. Z. A. El-Gamal, M. Elmogy, and A. Atwan, "Current trends in medical image registration and fusion," Egypt. Informatics J., vol. 17, no. 1, pp. 99–124, 2016, doi: 10.1016/j.eij.2015.09.002.

[22] D. P. Bavirisetti and R. Dhuli, "Fusion of Infrared and Visible Sensor Images Based on Anisotropic Diffusion and Karhunen-Loeve Transform," IEEE Sens. J., vol. 16, no. 1, pp. 203–209, 2016, doi: 10.1109/JSEN.2015.2478655.

[23]R. Srivastava, O. Prakash, and A. Khare, "Local energy-based multimodal medical image fusion in curvelet domain," IET Comput. Vis., vol. 10, no. 6, pp. 513–527, 2016, doi: 10.1049/iet-cvi.2015.0251.

[24] V. Bhateja, H. Patel, A. Krishn, A. Sahu, and A. Lay-Ekuakille, "Multimodal Medical Image Sensor Fusion Framework Using Cascade of Wavelet and Contourlet Transform Domains," IEEE Sens. J., vol. 15, no. 12, pp. 6783–6790, 2015, doi: 10.1109/JSEN.2015.2465935.

[25] S. V and B. R. Kumar, "Directive Contrast Based Multimodal Medical Image Fusion in NSCT with DWT Domain," Int. J. Eng. Trends Technol., vol. 9, no. 6, pp. 288–294, 2014, doi: 10.14445/22315381/ijett-v9p257.

[26] G. Bhatnagar, Q. M. J. Wu and Z. Liu, "Directive Contrast Based Multimodal Medical Image Fusion in NSCT Domain," in IEEE Transactions on Multimedia, vol. 15, no. 5, pp. 1014-1024, Aug. 2013, doi: 10.1109/TMM.2013.2244870.

[27]A. Mohammad-Djafari, "A Bayesian Approach for Data and Image Fusion," no. June, pp. 386–408, 2003, doi: 10.1063/1.1570554.

[28] I. Daubechies and W. Sweldens, "Factoring Wavelet Transforms into Lifting Steps," J. Fourier Anal. Appl., vol. 4, no. 3, 1998, doi: 10.1007/bf02476026.

[29] BRATS, 2015 https://smir.ch/BRATS/Start, accessed May 2019.



"Bodh", BPIT's International Journal of Technology & Management ISSN: 2454-8421, Volume 7, Issue 2, July- Dec 2021, Page 43-57

# Smart Secured Voting Tool Using Block-chain Technology in Peer-to-Peer Network

Bhawna Suri<sup>1</sup>, Shweta Taneja<sup>2</sup>, Rohan Sharma<sup>3</sup>, Muskaan Dua<sup>4</sup>, Rajneesh Dubey<sup>5</sup>

<sup>1</sup>Department of Computer Science and Engineering, Bhagwan Parshuram Institute of Technology, Delhi, India, <sup>1</sup>bhawnasuri@bpitindia.com, <sup>2</sup>shwetataneja@bpitindia.com, <sup>3</sup>rohan01101999@gmail.com, <sup>4</sup>muskaan.dua1999@gmail.com, <sup>5</sup>rkrajnees3@gmail.com

Abstract- Electronic Voting (e-voting) is the well-known method of casting and/or counting votes electronically. This is a cost-effective and efficient approach for handling a voting procedure, which has the feature of being benevolent as well as presenting data and soliciting high safety. Procuring e-voting is the need of the day and can be accomplished using only by the domains of communications and net-working. The major concerns are the data privacy for e-voting. We have proposed a tool- E-Matdan using block chain for safe and reliable of evoting. It is a three-step process, in step the mapping of a synchronized model of voting records based on Distributed Ledger Technology (DLT) is done to avoid fabrication of the votes. Secondly, on the using Elliptic Curve Cryptography (ECC), user credential model is made for validation of votes. Thirdly, a withdrawal model is made that allows the voters to change their vote before a preliminary to the decided deadline. By amalgamating the above constructions of different designs, we have proposed a block chain based e-voting scheme in P2P network. To showcase the work, using block chain an e-voting system is implemented on Linux platforms in the P2P network for sundry candidates.

Keywords--- Tensor, Gaussian, Decomposition, Corel-10K

#### I. INTRODUCTION

Democratic voting is the necessity of every Democratic country. The most familiar and easy voting technique was paper-based but now with so much advancement in technology, this casting of votes would save paper and minimize the cost of conducting elections. Another advantage of get ridding of this traditional paper based scheme with a new election system is to reduce forgery, booth capturing and other unfair means to a clean and fair elections [1].

Voting is the modus operandi to make a collaborative resolution or communicate an opinion among an assembly or a meeting or constituency [2]. Voting usually follows debates and discussions, controversies and election crusades. Since the 17th century, voting has been the accustomed technique by which contemporary exemplary democracy has set off. During voting, the individual to be elected is the one contesting the election, also known as the candidate of the election and the one who casts a referendum for their selected contestant is the voter. Customarily, the voter can vote in obedience to the list of candidate/contestant or vote for any other individual(s) he/she favors. Voting plebiscites must be anonymous and stamped by the voters in private kiosks so that no one else can discover out for [whom an individual is voting. Voting is also used in various other distinct privatized establishments and brackets, such as clubs, corporations, and volitional consortiums.

With the expeditious evolution of the Internet and information technologies, many run-of-themill offline amenities such as casting votes, sending/receiving mails, payments, are hiking up

to the online ones. Online voting is also well known as electronic voting (e-voting). E-Voting is an electronic means for registering and enumerating votes. E-voting helps us save time and endeavor with high productivity and pliability. Users of e-voting are balloters and election officials. The balloter/voter can proffer his/her votes in a computerized manner (electronically) to the election officialdom from any placement through the way of e-voting. The election administration is accountable for accumulating the votes from the balloters/voters. It is surely getting more and more recognition instead of the traditional method of voting. With the development of the Internet, e-voting became a crucial means of countless corporations.

#### A. E-Voting

Electronic voting machines (EVMs) are generally considered as weak machines for security concerns. Anyone who captures the machine or has access to it can interfere with it and can change the vote casts through that machine. Hence, there is a need to move one step ahead for this fair voting which is digital voting.

In digital voting the electronic devices, such as voting machines or an internet browser, are used to cast votes. The voting done through EVMs, which are at the polling stations, is also called e-voting. When the voting is done using a web browser is known as i-voting. Every mechanism has their pros and cons, the main concern in i-voting is the security issues. To handle the security the safest track is blockchain technology.

Blockchain technology was primarily designed for cryptocurrency- bitcoin. It uses the distributed architecture, and every transaction is stored as a block chain. It is a secure and robust system and can be used for digital voting.

#### B. Role of Blockchain in E-voting

A blockchain is a distributed, immutable, incontrovertible, public ledger with 4 main features – storage at many different locations, no single point of failure in the maintenance, every new block holds the reference the previous version of the ledger, creating an immutable chain and lastly every new block becomes a permanent part of the ledger. The first block in the blockchain is known as the 'Genesis block' or 'Block 0'. [3] All this is based on the cryptography techniques, providing a more secure database than the previous databases. The blockchain technology can therefore be considered as the ideal tool for the voting process.

The design of blockchain in e-voting is based on DLT which is a list of blocks It is represented as a series of voting blocks chained sequentially chained to each other. The first block is known as the Genesis block. This is shown in Fig 1.



©"BBIJTM 2022", All Rights ReservedPage 48

#### Fig. 1: E-Voting blockchain

In each block, there is a voter's ID, voter's signature, timestamp, vote and a digest (hash) as shown in fig 2.



Fig. 2: A voting block

(1) Voter's ID: Voter is the person who casts a vote to his/her chosen candidate. ID is assigned randomly to the person who has the right to vote.

(2) Vote: A ballot is assigned to the chosen candidate of the voter.

(3) Voter's signature: A voter uses his/her private key to assign the vote in an encrypted manner. This encryption is done with the help of private key and hash generation. No one else can find out to whom the ballot is assigned.

(4) Timestamp: Submission of ballot is tracked by a timestamp. If two or more blocks have the same timestamp, the one with higher value of signature is preferred over others.

(5) Hash of the previous block: SHA-256 algorithm is used to compute the previous block hash value.

Thus, the blockchain-based e-voting scheme is immune to data modification and is non-repudiation.

## C. Our Contribution

In our work, blockchain technology is used to improve the security of e-voting. Following are our contributions:

(1) A synchronized model of voting records has been designed based on DLT to avoid forgery of votes.

(2) A user credential model has been designed based on ECC to provide authentication and non-repudiation.

(3) A withdrawal model has been developed that enables voters to change their vote before a preset deadline.

By integrating the above designs, we propose a blockchain-based e-voting scheme, which meets the essential requirements of the e-voting process.

Following are the features of our blockchain-based e-voting scheme as follows:

(1) The blockchain-based e-voting scheme is public, distributed, and decentralized. Votes can be recorded from voters across the world using mobile or computers

(2) In blockchain-based e-voting schemes voters are allowed to audit and verify the votes inexpensively.

(3) The management of the database of votes is done autonomously and uses a distributed server of the timestamp on a peer-to-peer network.

(4) Process of voting on the blockchain is a workflow where voters regarding data security are marginal, which removes the characteristic of infinite reproducibility from e-voting.

The paper is organized as follows. Section1 deals with introduction. Section 2 deals with the related work. Problem formulation is described in the Section 3, Section 4 and its subsection deals with the estimation of the context information. Mathematical model for context estimation comes in Section 5. Section 6 deals with result and discussion .Section 7 deals with conclusion and future scope. Section 8 contains the reference part.

#### II. RELATED WORK

Block chain technology can be applied in many applications, here we have discussed its applications done in different manners and in different domains. Anonymous voting by two-round public discussion, proposed an addition of a self-tallying function to the 2-Round Anonymous Veto Proto-col (called AV-net). The AV-net provided exceptional efficiency compared to related techniques, the paper was focused on the dining cryptographers network (DC-net) and its weaknesses and proposed the AV-net as a new way to tackle that problem [4]. The new protocol, like the AV-net requires no trusted third party or private channel. Participants execute the protocol by sending two-round public messages, but are significantly more efficient in terms of the number of rounds, computational cost and bandwidth usage.

In general, the new protocol divided electronic voting into to two classes:

- 1) Decentralized elections where the protocol is essentially run by the voters.
- 2) Centralized elections where trusted authorities are employed to administer the process.

The goal there-fore was to eliminate the use of a trusted third party altogether. The first round in the two-round protocol consisted of every participant to publish his public key and a zero knowledge proof (ZKP) for his private key. When the round finished, each participant checks the validity of the ZKPs and computes. In the second round, each participant needs to demonstrate that the encrypted vote was one of the valid voting choices without revealing which one. The authors in [5] have proposed the first implementation of a decentralized and self-tallying internet voting protocol with maximum voter privacy using the Block chain, called The Open Vote Network (OVN). The OVN is written as a smart contract for the Ethereum block chain. It is costing 0.73\$ per voter. In this system the voting is conducted in an unsupervised environment. The OVN is also vulnerable to denial-of-service The implementation is feasible only for small boardroom voting, with the disadvantage that each voter has to download the full Ethereum block chain to confirm the voting protocol is being executed correctly.

A multi authority secret-ballot election scheme which would guarantee privacy, universal verifiability and robustness, where voters would participate using a PC, where the main consideration is the effort required of a voter [8]. In this model, voters cast their vote by posting ballots to a bulletin board. The bulletin board works as a broadcast channel with memory to the extent that any party can access its content but no party can erase anything from the bulletin board. The ballot does not reveal any information on the vote itself but is ensured by an accompanying proof that the ballot contains a valid vote. The final tally, the sum of all votes,

which occurs when the deadline is reached, can then be obtained and verified, by any observer, against the product of all submitted ballots. Which would ensure universal verifiability, due to the homomorphic properties of the encryption method used? While this proposal can scale up to large elections better than the previous ones, it does have limitations.

Netvote[9] is a decentralized block chain-based voting network on the Ethereum block-chain. Netvote utilizes decentralized apps for the user interface of the system. The Admin dApp allows election administrators to set election policies, create ballots, establish registration rules and open and close voting. The Voter dApp is used by individual voters for registration, voting and can be integrated with other devices (such as biometric readers) for voter identification. The Tally dApp is then used to tally and verify election results.

Netvote supports the three main types of elections as:

- 1) Open Election: Anyone may vote
- 2) Private Election: Only authenticated and authorized individuals may vote
- 3) Token-Holder Elections:

Another approach that only voters who operate accounts that have a balance of a designated compliant token may vote is implemented on Ethereum network. Here the people who do not even have an Ethereum wallet are allowed to vote. Users can give votes through their Android device or directly from their Ethereum wallets, and these transaction requests are handled with the consensus of every single Ethereum node for e-voting [10]. In [11], the Block-chainenabled e-voting (BEV) is implemented to reduce voter frauds. Eligible voters can cast their vote through a computer or smartphone. BEV uses an encrypted key and tamper-proof personal IDs for e-voting. Block chain is offering new opportunities to develop new types of digital

services. In [12], block chain technology is used for electronic voting that could be used in local or national elections and help to increase the trust of voters as well as governments.

#### III. PROPOSED METHODOLOGY

The objective of this paper is to develop a safe and secure tool for voting over the internet and this tool is E-Matdaan. The step by step execution of the framework of the E-voting is, Firstly the user registers for an electronic ID, a user chooses a PIN number for its corresponding ID consisting of 6 numbers. A user will therefore identify himself in the voting booth by scanning his ID and providing his corresponding PIN number to authenticate himself to the system.

Using the following methods:

1) Any computer in any voting district can be used by any eligible voter to vote, by checking its authentication. As the wallet for the corresponding voter has information about the voter.

2) On successful authentication, the corresponding smart contract which is a ballot is prompted for choosing the candidate to which the user wishes to vote.

3) Voter now must re-enter the corresponding PIN number for his electronic ID for casting vote.

4) Once the vote is signed the data is verified by the district node. If the node accepts the vote data, the vote data must be agreed upon by the majority corresponding district node.

5) If the majority of district nodes agree upon the vote data, consensus for the particular vote has been reached. The user then receives the transaction ID for the corresponding transaction of his vote in the form of a QR-code and the option to print the transaction ID. After this, smart contract adds one vote to the party for which the user has voted for. This functionality is used for voting and displaying results at district level.

6) Now all the received and verified transactions are added as blocks in the block chain after the threshold time is reached. The block chain is then updated with the entry of every new block and each district node updates this copy of the ledger as in fig3.



Fig. 3: E- voting proposed Methodology

```
©"BBIJTM 2022", All Rights ReservedPage 56
```

#### IV. IMPLEMENTATION AND RESULTS

This section illustrates the design and functional phase of our application E-Matdan are shown in Fig. 4, 5, 6 and 7.

The different phases of the tool are discussed below:

1. **Registration Phase:** The Voter has to register itself first with its unique id and attributes such as name and mobile number. All this data is stored in the database.

2. **Login:** The voter after registration tries to login using password. After successful login, to cast their vote voter has to authenticate themselves using OTP authenticity.

3. **Blockchain Technology**: Blockchain encrypts the vote casted using Asymmetric encryption algo-rithm. A public key is provided by Blockchain, and private key is with the host. Public key is used for verification purposes by the ledger.

5. **Ethereum Network:** Ethereum network provides a framework for blockchain creation and stor-age. Every block is created and its details are stored in an encrypted ledger using Solidity. These created blocks are distributed among nodes which provide high fault tolerance to the system.



Fig. 4: Home Page of E-Matdaan





Fig. 6: Admin section of E-Matdaan

Fig. 7: Results Section of E-Matdaan

## V. CONCLUSION AND FUTURE DIRECTIONS

In this research paper a smart and secured tool - E-Matdaan for safe and secure voting using block-chain has been proposed. The block chain has been implemented using the Ethereum network. The idea of adapting digital voting systems to make the public electoral process cheaper, faster and easier, is a compelling one in modern society. Making the electoral process cheap and quick, normalizes it in the eyes of the voters, removes a certain power barrier between the voter and the elected official and puts a certain amount of pressure on the elected official. It also opens the door for a more direct form of democracy, allowing voters to express their will on individual bills and propositions.

We have outlined the systems architecture, the design, and a security analysis of the system. By comparison to previous work, we have shown that blockchain technology offers a new possibility for democratic countries to advance from the pen and paper election scheme, to a more cost- and time-efficient election scheme, while increasing the security measures of today's scheme and offering new possibilities of transparency. Using an Ethereum private block chain, it is possible to send hundreds of transactions per second onto the block chain, utilizing every aspect of the smart contract to ease the load on the block chain. For countries of greater size, some measures must be taken to withhold greater throughput of transactions per second, for example the parent & child architecture [28] which reduces the number of transactions stored on the block chain at 1:100 ratio without com-promising the network's security. Our election scheme allows individual voters to vote at a voting district of their choosing while guaranteeing that each individual voters vote is counted from the correct district, which could potentially increase voter turnout

#### REFERENCES

[1].Van der Elst, C., & Lafarre, A. (2017). Bringing the AGM to the 21st century: Blockchain and smart contracting tech for shareholder involvement. European Corporate Governance Institute (ECGI)-Law working Paper, (358).

[2].Jonéus, C. (2017). Analysis of Scalable Blockchain Technology in the Capital Market.

[3].Gupta, A., Patel, J., Gupta, M., & Gupta, H. (2017). Issues and Effectiveness of Blockchain Technology on Digital Voting. International Journal of Engineering and Manufacturing Science, 7(1), 20-21.

[4].Henning, J., & Schreiber, R. (2014). The Bitcoin Protocol. Technische Berichte des Hasso-Plattner-Instituts für Softwaresystemtechnik an der Universität Potsdam, 18.

[5].Saad, M., Spaulding, J., Njilla, L., Kamhoua, C., Shetty, S., Nyang, D., & Mohaisen, D. (2020), Exploring the attack surface of blockchain: A comprehensive survey. IEEE Communications Surveys & Tutorials, 22(3), 1977-2008.

[6].Lazarenko, A., & Avdoshin, S. (2018, November). Financial risks of the blockchain industry: A survey of cyberattacks. In Proceedings of the Future Technologies Conference (pp. 368-384). Springer, Cham.

[7].Springall, D., Finkenauer, T., Durumeric, Z., Kitcat, J., Hursti, H., MacAlpine, M., & Halderman, J. A. (2014, November). Security analysis of the Estonian internet voting system. In Proceedings of the 2014 ACM SIGSAC Conference on Computer and Communications Security (pp. 703-715).

[8].Eze, P., Eziokwu, T., & Okpara, C. (2017). A triplicate smart contract model using blockchain technology. Circulation in Computer Science-Special Issue, 1-10.

[9].Mus, K., Kiraz, M. S., Cenk, M., & Sertkaya, I. (2016). Estonian voting verification mechanism revisited. CoRR, abs, 1612.

[10].E. Yavuz, A. K. Koç, U. C. Çabuk and G. Dalkılıç, "Towards secure e-voting using ethereum block-chain," 2018 6th International Symposium on Digital Forensic and Security (ISDFS), 2018, pp. 1-7, doi: 10.1109/ISDFS.2018.8355340.

[11].N. Kshetri and J. Voas, "Blockchain-Enabled E-Voting," in IEEE Software, vol. 35, no. 4, pp. 95-99, Ju-ly/August 2018, doi: 10.1109/MS.2018.2801546.

[12]. Ayed, A. B. (2017). A Conceptual Secure Block chain-based Electronic Voting System. International Journal of Network Security & Its Applications, 9(3), 01-09.



"Bodh", BPIT's International Journal of Technology & Management ISSN: 2454-8421, Volume 7, Issue 2, July- Dec 2021, Page 58-69

## Smart Agricultural Robot Usha Sharma<sup>1</sup>, Megha Sharma<sup>2</sup>, Rupal Garg<sup>3</sup>

<sup>1</sup>Department of Electronics and Communication Engineering, Bhagwan Parshuram Institute of Technology, Delhi, India, <sup>1</sup>usha1984sharma@gmail.com, <sup>2</sup>meghashrm286@gmail.com, <sup>3</sup>rupalgarg3@yahoo.in

Abstract: This paper represents technological enhancement in the field of agriculture. Agriculture is the most important part of the country's economy as it gives great contribution to the development of the country. Here we are proposing a wireless agricultural robot which will perform basic operations done in farming activities so that it will reduce the labor size, time and energy required to perform the different tasks. The concept of smart agricultural robot represents a Wi-Fi controlled robot that will perform digging mechanism, seed sowing mechanism and water irrigation after sensing the temperature and moisture of soil. Moreover, the proposed mechanism gives a solution for farmers to automate irrigation. Farmers may make decisions based on monitoring different parameters like humidity, soil moisture, temperature etc received on his mobile app.

Keywords--- Agriculture, Robot, GPS, GSM, ARM7

#### I. INTRODUCTION

Few years back farmers were using bullock to perform different agricultural task, after some advancement in the technology farmers are now using tractors for seed sowing operations but still there are some tasks which need to be performed by farmers manually, so to reduce man power and efforts we are designing a smart agricultural robot which will be controlled by farmers using wireless technology, and it will perform basic farming operations. Many systems were developed in past years to provide the best solutions in a field of agriculture which results in greater yield of different crops. Nowadays technology is becoming a great contributor in this field as with its advancement systems are developed using microcontrollers to perform different tasks like surveillance, fruit picking and other operations as well. A precision agriculture robot for seeding function was being researched in which a robot was developed which helps in sowing seed to specific depth according to crop requirement [1]. A Smart Farming System Using Sensors for Agricultural Task Automation was developed which works with different sensors to predict the moisture and nutrient contents of soil and works as water irrigator [2].

A Smart farming using IOT system is being researched which works with obstacle sensor and other sensors and performs cutting and spraying functions [3]. A GPS based autonomous agricultural robot was developed which uses GPS and magnetometer and performs plowing, leveling, and gives message indications to start the irrigation process [4]. By integration of hardware and software automated devices can carry out various processes like seed sowing, irrigation, sensing of temperature, humidity and fertilization. Various Wireless networks are

utilized in respect to control the action from the remote place. It may be done using GSM, Sensors network, Zigbee, Bluetooth and IOT [5-8].

Parameters	Precision robot	Smart Farming using Sensors	Smart Agricultural robot using IOT
Microcontroller	ARM7	Ardiuno	AVR
Technology used	GSM based	IOT based	Telnet based
Working	Precise seed sowing operation	Sensing moisture and perform irrigation	Sense temperature, moisture, distance and perform basic operations
Cost	Expensive	Low cost but limited operations	Cost effective
Operations	Perform seed sowing only	Performs irrigation only	Performs irrigation seed sowing and digging.

Table I: Comparison table to summarize the differences with other systems

The proposed architecture uses the Wi-Fi module to send the sensed data from sensors to farmer mobile app. Farmer can also give commands from his phone to the robot present at his fields to perform the needed activity. A summary of various parameters which are helpful in opting the solution among the present are listed in comparison Table I .This paper will work on different parameters like temperature and soil moisture content according to crop requirement. The main objective is to build a general purpose robot that will help in performing different

operations related to farming. This paper also aims at using cost-effective technology which will be affordable and easily accessible to the farmers. The paper is organized in three main subsections. Section II describes the architecture of the proposed robot. In section III the working of the proposed prototype will be discussed. Section IV will investigate the final hardware Robot and the results.

#### II. ARCHITECTURE OF PROPOSED ROBOT

In past years many systems were developed which were limited to performing one or two tasks like seeding, water irrigation and weeding. Those systems were based on wireless technologies like GPS which makes the system expensive and prohibitive for farmers. This paper strives at developing a system using inexpensive and lightweight components which makes it low-end and compact. The aim is to provide a cost-effective system for farmers and to reduce labor, manpower, noise and it helps in reducing pollution created by tractors these days. Fig.1 depicts the block diagram for the smart agricultural robot which senses the temperature and soil moisture of a field and performs other agricultural tasks like digging, seed sowing and water irrigation using mobile applications with the help of Wi-Fi module.



Fig.1: Block Diagram for the proposed smart agriculture system

## **III. METHODOLOGY**

In agriculture, parameters like temperature and soil moisture are important for crop production as these parameters will detect whether a particular crop can be grown in that area or not. The

temperature of field is observed using a LM35 temperature sensor, in accordance with this water content of soil is sensed by soil moisture sensor and an ultrasonic sensor is used to detect obstacle in the fields and also detects the fencing of field 50cm beforehand, all these conditions are displayed on  $16\times2$  LCD. When conditions of field are perfect for a particular crop then user can send command to the robot using a mobile application which is connected with the Robot through a Wi-Fi module named as ESP8266, this module will receive commands from mobile application and send it to AVR microcontroller then it will send these commands to other components to perform a desired task.



Fig.2. Screenshot of Mobile application

Fig.2 shows the screenshot of a mobile application used by the farmer to control the proposed smart agricultural robot.

Operations	Task performed
Forward#	Moves robot in forward direction
Backward#	Moves robot in backward direction
Right#	Moves robot in right direction
Left#	Moves robot in left direction
Sow#	Rotates the seed bottle to drop seeds
Up#	Moves digger upward
Down#	Moves digger downward
Pump on#	On water pump
Pump off#	Off water pump
Send status#	Send temperature, moisture content, distance to app
Stop#	Stops movement of robot

#### Table II: List of Operations given on mobile app

Table II. Elucidates the operations performed on pressing different keys on mobile applications. By a single click on the operation tab, users can control the robot at fields to perform the appropriate activity to yield the crop. The input is given to ATmega328 IC through the Wi-Fi module and output is obtained at the mechanical parts of the robot. For moving the robot forward or backward DC motors are connected with wheels which will move the robot easily on normal as well as rugged fields. For digging mechanism, a clipper is connected with the help of a dc motor which will dig soil when a command to perform digging operation is received using Wi-Fi module. To perform seed sowing operation a bottle with holes in it is connected to the system using a DC motor so that seeds can be dropped in soil easily by rotating the bottle using a DC motor. When the user sends a command for seeding operation from a mobile application the microcontroller will receive the command and send it to the DC motor which is connected to the seed bottle and make it to drop seeds in soil. In order to make soil wet or perform water irrigation a submersible DC water pump is connected in a system which will receive command through Wi-Fi to plunge water in soil which will help in keeping moisture content of soil as required. An ultrasonic sensor is added in the system which detects fencing of the field before 50cm of reaching the fencing so that when sensor detects the fencing it will stop and send message to user that obstacle is detected on mobile app of user.



Fig.3 Final proposed prototype with LCD display

Fig3 shows an image of the final robot which will perform basic agricultural tasks and a LCD display to show the reading taken up while the experiment has been performed. A small piece of land is used to perform the experimental setup to test various operations as listed in Table3.

## IV. RESULTS AND DISCUSSION

The main aim of developing a multipurpose system is to reduce time and energy consumed in performing agricultural tasks. This system also aims at making a cost-effective aid which can replace existing systems and help in efficient crop production. This robot will perform basic operations of agriculture easily and it will be beneficial for farmers to get solutions to their problems in hand. To check the performance of the robot, experiments are done on both dry soil and wet soil and moisture content is displayed on display for both soils.



Fig.4. LCD output for the conditions of Dry soil



Fig.5. LCD output for the conditions of Wet soil

When experiments were performed the time required for performing different operations have been calculated. Table 3 presents the time delays during the various operations performed. This

system will perform basic agricultural tasks in a very less time and help in reducing labor and manpower.

Operations	Task performed	Time delay	
Connection power supply	Supply power to whole circuit	2-3 secs	
Wi-Fi connection	Connects robot to internet	5-6 min	
Movement			
Forward	Moves robot forward	2-3 secs	
Backward	Moves robot backward	2-3 secs	
Left	Moves robot left	2-3 secs	
Right	Moves robot right	2-3 secs	
Irrigation			
Pump On	On water pump	2-3 secs	
Pump Off	Off water pump	2-3 secs	
Sowing			
Sow	Rotates the seed bottle to drop seeds	2-3 secs	
Digging			
Up	Moves digger upward	2-3 secs	
Down	Moves digger downward	2-3 secs	
Send status	Send parameters details to app	4-5 secs	
Stop	Stops movement of robot	2-3 secs	

	Table 3:	Calculated	time	delays
--	----------	------------	------	--------

#### VIII. CONCLUSION AND FUTURE SCOPE

A smart agricultural robot is developed to perform basic tasks of agriculture i.e. sensing temperature, sensing moisture content of soil, digging, seed sowing and water irrigation. This robot is developed using low end and accessible components which makes it cost effective. In ©"BBIJTM 2022", All Rights ReservedPage 72
order to sense temperature LM35 temperature sensor is used and for observing the moisture content in soil, moisture sensor is used which helps in providing greater yield of different crops. Technology used in this system is easy to understand which makes it user friendly. In this system Wi-Fi module is used for serial communication between AVR microcontroller and mobile application which is easily accessible. The design of the robot is less complex and manageable which proves to be a quick fix to farmer's problems. This robot helps in reducing man power, labor, cost, noise and it helps in decreasing pollution created by tractors these days. This robot can be a good choice over tractors which are being used now as it is an environment friendly system. For further advancements we can use solar panels to provide power supply to the system which makes the system less dependent on electricity because there are large amounts of power cuts in rural areas. We can also add different operations in the robot like spraying pesticides, cutting weeds etc.

#### REFERENCES

- [1].N. S. Naik, V. V. Shete and S. R. Danve, "Precision agriculture robot for seeding function," 2016 International Conference on Inventive Computation Technologies (ICICT), 2016, pp. 1-3, doi: 10.1109/INVENTIVE.2016.7824880.
- [2]. Chetan Dwarkani M, Ganesh Ram R, Jagannathan S and R. Priyatharshini, "Smart farming system using sensors for agricultural task automation," 2015 IEEE Technological Innovation in ICT for Agriculture and Rural Development (TIAR), 2015, pp. 49-53, doi: 10.1109/TIAR.2015.7358530.

- [3]. Vadapalli, Adithya & Peravali, Swapna & Dadi, Venkatarao. "Smart Agriculture System using IoT Technology" 2020 International Journal of Advance Research in Science and Engineering (2319-8354). Vol 09. Pp 58-65.
- [4].K. Shaik, E. Prajwal, S. B., M. Bonu and B. V. Reddy, "GPS Based Autonomous Agricultural Robot," 2018 International Conference on Design Innovations for 3Cs Compute Communicate Control (ICDI3C), 2018, pp. 100-105, doi: 10.1109/ICDI3C.2018.00030.
- [5].T. Choudhury, A. Kaur and U. S. Verma, "Agricultural aid to seed cultivation: An Agribot," 2016 International Conference on Computing, Communication and Automation (ICCCA), 2016, pp. 993-998, doi: 10.1109/CCAA.2016.7813860.
- [6].S. Jaiganesh, K. Gunaseelan and V. Ellappan, "IOT agriculture to improve food and farming technology," 2017 Conference on Emerging Devices and Smart Systems (ICEDSS), 2017, pp. 260-266, doi: 10.1109/ICEDSS.2017.8073690.
- [7].S. A. Amrita, E. Abirami, A. Ankita, R. Praveena and R. Srimeena, "Agricultural Robot for automatic ploughing and seeding," 2015 IEEE Technological Innovation in ICT for Agriculture and Rural Development (TIAR), 2015, pp. 17-23, doi: 10.1109/TIAR.2015.7358525.
- [8]. Vishnu Prakash K, Sathish Kumar V, Venkatesh P, Chandran A, Design and fabrication of multipurpose agricultural robot, International Journal of Advanced Science and Engineering Research, Volume: 1, Issue: 1,pp 23-25, June 2016, ISSN: 2455 9288.



"Bodh", BPIT's International Journal of Technology & Management ISSN: 2454-8421, Volume 7, Issue 2, July- Dec 2021, Page 70-88

# Camera Model Identification with Convolution Neural Network

#### Harsh Kaushik<sup>1</sup>, Harshita Chadha<sup>2</sup>, Neelam Sharma<sup>3</sup>, Harsh Nigam<sup>4</sup>, Eklavya Sharma<sup>5</sup>, Komal Bhagat<sup>6</sup>

 <sup>1, 2, 3</sup>Department of Computer Science & Engineering, Maharaja Agrasen Institute of Technology, New Delhi, India
<sup>4, 5, 6</sup>Department of Electronics & Communication Engineering, Bhagwan Parshuram Institute of Technology Delhi, India
<sup>4</sup>harsh.nigm@gmail.com, <sup>5</sup>ekyusharma@gmail.com, <sup>6</sup>bhagat.k15@gmail.com

Abstract—With the vast improvements in multimedia, image forgery has become a growing threat to forensics. In this research, we present a deep learning-based strategy for tracing an image back to its original camera model. We categorized a set of seven camera models; however the concepts presented in this research can be easily expanded to classification of almost any number of models based on the training data provided to the system. A comparison of dense neural networks and convolutional neural networks was conducted, and both of these cutting-edge techniques produced a robust solution.

Keywords—Dense Neural Networks, Convolutional Neural Networks. Digital Images, Camera model identification.

#### I. INTRODUCTION

With the advent of technology and the recent influx of gadgets, cyber forensics has become a hotbed of computer related research. In today's technology oriented world, there is no shortage of digital footprints and these can inadvertently help in solving criminal cases, finding individuals, carrying out relevant research, etc. [1]. For more than a decade, identifying the brand and model of the camera that recorded a picture has been a significant study subject in information forensics [2]. Determining the type of camera used to acquire an image can aid in determining its source. Although metadata may convey information about the origin of a picture, it is simple to change [3]. As a result, signal processing is developed to determine the inherent traces left on a picture after the digital cycle. This has applications in proof verification because it substantially reduces the time required by using digital technology; an algorithm to fix this can be placed on smartphones, enabling real-time verification while eliminating the logistics of transporting image/video proof to a forensic lab [4]. It will be far more crucial in a country like India due to the absence of judges and an increase in case time. Up until now, hand-crafted filtering and feature extraction techniques have largely been used in camera source identification and other image forensics [5]. However, following the success of picture recognition reported in a number of deep learning based studies have recently appeared. The camera model problem illustration is given in Fig. 1. To create an algorithm, we must first comprehend the process of capturing a digital image. Fig. 2 depicts the pipeline and several steps of the picture capture process. In reality, current forensic research has proven that each physical and algorithmic component creates an inherent trail that is individually identifiable to the camera model, and these remnants may be employed to connect the image to the model of its manufacturer [6].



Fig.1. Camera Model Problem Illustration

Various camera types, for instance, employ different lenses, which might inject distinct noise into the image. The camera gathering process pipeline is given in Fig. 2. According to recent forensic research, each physical and algorithmic component generates an inherent trace that is uniquely traceable to the camera model [7]; those remnants may be utilized to link the image to the model of its manufacturer. For instance, various camera types employ different approximation demo-spacing methods.

©"BBIJTM 2022", All Rights ReservedPage 78



Fig.2. Camera gathering process pipeline

In this research article, we present a convolution neural network based solution for the posited problem. We first articulate the problem in section 1. In section 2, discuss notable work done in the recent past that has been conducted in this field. Sections 3-5 are dedicated towards a description of the presented proposed approach. Section 6, presents the results followed by Conclusion and Future research directions.

## II. LITERATURE REVIEW

In previous works, rule based approaches have been explored. Feature engineering is done by domain experts and despite the fact that these methods have produced very good results but they are limited to their respective models and data, their extension is not guaranteed to work on unknown camera models as they don't have the ability to adapt to new data sources [8]. Statistical models have been developed to model the sensor noise pattern to classify the model of the camera used [7], [8]. Features based on JPEG compression have been built to solve this challenge [9].

For a brief overview of the research in this field, survey papers can be referred to [10]–[13]. Hu et al. [14] suggested three types of local characteristics for pedestrian portrayal: Hierarchical Weighted Histograms (HWH), Gabor Ternary Pattern HSV (GTP-HSV), and Maximally Stable Color Regions (MSCR), which could impose limiting factors on the part-level, pixel-level, and blob-level, including both. Based on the symmetry and asymmetry perceptual concepts, Bazzani et al. [15] proposed the Symmetry-Driven Accumulation of Local Characteristics (SDALFs) to identify three complementary characteristics for pedestrian photos. Yang et al. [16] created the Salient Color Names-based Color Descriptor (SCNCD) to compute colour name patterns across multiple colour models in order to manage lighting fluctuations. The Local Maximal Occurrence (LOMO) [17] technique seeks to overcome viewpoint variations by maximizing the frequency of extracted features of horizontal areas. There was a time when hand-crafted features were popular. [18]–[19] Prior to the widespread adoption of deep learning technologies in humans Gheissari et al. [20] used normalized colour and salient edge histograms to depict pedestrian pictures that are resistant to changes in surroundings and aspects.

# III. ARTIFICIAL NEURAL NETWORKS

Artificial neural networks, the architecture of which is influenced by the biological brain, [21] have been known for a long time and have historically been able to tackle certain types of issues. However, more recently, neural network architectures that contained layers of hidden neurons (in addition to the input and output layers) were developed, and it is this added degree

of complexity that enables deep learning and gives a more powerful set of problem-solving capabilities [22-23]. The general model of the deep neural network is given in Fig3.



Fig.3. Deep Neural Network model

Since ANNs have such a wide range of architectures, there is no universally accepted neural network description. The ability to approximate nonlinear functions and the possession of adaptive weight sets are two common aspects of all ANNs [24].

# IV. CONVOLUTION NEURAL NETWORKS

Convolution Neural Networks are a class of deep learning algorithms that are based on the mathematical operation of convolution [25]. These powerful neural networks that derive

inspiration from the natural world form the backbone of the computer vision industry. A layered neuron architecture is employed here and the spatial characteristics of the images remain intact which is one of the main driving forces behind their large-scale success [26]. The general model of a convolution neural network is given in Fig. 4.



Fig.4. A general Model of a Convolution Neural Network

In keeping with the general rule of thumb when it comes to convolutional neural networks, the first few layers are charged with detecting low-level features and the deeper the network is the more details are detected. The matrices are called as kernels are convolved, i.e., a weighted sum of products is taken, with the image [27]. This exercise is repeated multiple times often coupled

with pooling operations to reduce image dimensionality. Some combination of dense/ fully-connected layers is also often used to derive meaningful insights [28].

# V. EXPERIMENTAL SETUP

To verify our model's efficiency and generalization capacity, we will partition the dataset into two sets: training and validation. The training set is used to train our model so that the CNN may learn high-level characteristics from it. The test set and train set do not intersect since doing so would result in the leaking of validation set information into training, rendering the split ineffective. Our issue is primarily one of categorization. We will utilize the Dresden database, which is used to create benchmarks for numerous image related challenges in the field of forensics, making comparison simple.

To stay below our computational budget, we will create our model for 7 distinct camera models, making it a 7 multi classification task. The resolution of each image was modified to 100\*100\*3. In training and testing, around 13000 photos were used.

# A. DENSE NEURAL NETWORKS (DNN)

The first approach employed is a basic dense connected neural network with a softmax function as the last layer. Cross entropy loss is the definition of loss, while accuracy is the assessment metric. We also look at the confusion matrix to observe how the DNN performs with various models. The model is based on local Fourier characteristics and a local binary pattern.

These characteristics capture the local variance in the image. We acquire roughly 250 features for each image once we extract these features; hence our training set has the shape (10000\*250).

The network's weights are randomly initialized, and RMS prop is used to update gradients. At highly linked layers, activation ReLU is utilized, whereas in the softmax layer, sigmoid is used. It took a day for the training to come together. The accuracy of 0.82 has been achieved on the test set. The model's parameters considered are as in Table1.

	8
LAYERS	SHAPE
input 1 (input	(None, 250)
layer)	
Dense 1 (dense)	(None,
	1000)
Dense 2 (dense)	(None, 512)
Dense 3 (dense)	(None, 512)
Dense 4 (dense)	(None, 256)
Dense 5 (dense)	(None, 128)
softmax1(dense)	(None, 7)

## Table.1. Summary of the model generated

## **B. CONVOLUTION NEURAL NETWORK (CNN)**

The second strategy is based on CNN, a specific neural network design for image-based challenges. Convnets, which have demonstrated effectiveness in recognition and classification, will be used in this design. Back-propagation is used to update the weights, the Adam optimizer to update the gradients, and the leaky-ReLU activation function in convolution layers and dense

layers. Convolution layers are designed to extract features, and as we go deeper, more advanced features can be extracted, and finally convolution layer, the output is flattened and fed to a dense layer, after that we have another 2 dense layers and then in the final layers which is the softmax layer, we make calculate the probabilities associated with each class for a particular instance.



Fig.5. Flowchart of DNN training

Training took the same time as above; we made sure the amount of parameters in this is the same as the previous approach to see which architecture makes better use of the parameters.

The flowchart of the architecture of DNN and CNN training model are as in Fig.5 and Fig.6. An Accuracy of 0.91 has been achieved for the test set.

	0	1	2	3	4	5	6
)	0.82	0.07	0.05	0.01	0.04	0.01	0
I.	0.07	0.83	0.02	0.03	0.01	0	0.04
2	0.05	0.02	0.84	0.08	0.01	0	0
3	0.01	0.03	0.08	0.79	0	0.08	0.02
4	0.04	0.01	0.01	0	0.84	0.07	0.03
5	0.01	0	0	0.08	0.07	0.789	0.054
6	0	0.04	0	0.02	0.03	0.054	0.832



©"BBIJTM 2022", All Rights ReservedPage 87

#### Fig.6. Flowchart of the CNN training process

## VI. RESULTS

Even though we didn't have a huge dataset and access to high-end PC, but from the above results, we can infer that CNN performs better than a simple DNN, as CNN is able to model spatial features we can assume that a camera model leaves particular kind of spatial noise in the image which can be modeled mathematically.

LAYERS	SHAPE
input1(InputLayer)	(None, 100, 100, 3)
batchnormalization1	(None, 100, 100, 3)
conv2d1(Conv2D)	(None, 100,100, 16)
batchnormalization1	(None, 100, 100, 3)
conv2d1(Conv2D)	(None, 98, 98, 16)
maxpooling2d1	(None, 46, 46, 16)
conv2d1(Conv2D)	(None, 46, 46, 32)

Table2. Summary of the model generated

batchnormalization1	(None, 46, 46, 32)
conv2d1(Conv2D)	(None, 42, 42, 32)
maxpooling2d1MaxPooling 2	(None, 16)
Dense2(dense)	(None,512)
Dense3(dense)	(None,512)
softmax1(dense)	(None, 7)

A dense neural network based and a convolution neural network based methods yielded accuracy values of 82% and 91% respectively.

Table.4. Confusion Matrix for DNN

	0	1	2	3	4	5	6
0	0.94	0.02	0.01	0.02	0	0	0.01
1	0.02	0.92	0.02	0.01	0.03	0	0
2	0.01	0.02	0.82	0.05	0.02	0.03	0.05
3	0.02	0.01	0.05	0.85	0.04	0.03	0
4	0	0.03	0.02	0.04	0.91	0	0
5	0	0	0.03	0.03	0	0.92	0.02
6	0.1	0	0.05	0	0	0.02	0.92

# VII. CONCLUSION AND FUTURE SCOPE

This paper has proposed a robust methodology to trace back an image to the camera model that was used to capture it. A dense neural network based and a convolution neural network based methodologies have been employed. Both methods yielded useful results with accuracy values

of 82% and 91% respectively. Variations of CNN and a much deeper network with skip connection can be used to achieve higher accuracy. We can also use pre trained networks like the Inception module, GoogleNet to harness the benefits of transfer learning. An interesting route would be to apply Capsule nets to model spatial relationships more concretely.

#### REFERENCES

- [1].M. C. Stamm, M. Wu, and K. J. R. Liu, "Information forensics: An overview of the first decade," IEEE Access, vol. 1, pp. 167–200, May 2013.
- [2].A. Piva. (2013). An overview on image forensics. ISRN Signal Process. [Online]. Available: https://
- [3].www.hindawi.com/journals/isrn/2013/496701/
- [4].M. Kirchner and T. Gloe, "Forensic camera model identification," in Handbook of Digital Forensics of Multimedia Data and Devices. Hoboken, NJ: Wiley, 2015, pp. 329– 374.
- [5].] A. Swaminathan, M. Wu, and K. J. R. Liu, "Digital image forensics via intrinsic fingerprints," IEEE Trans. Inf. Forensics Security, vol. 3, no. 1, pp. 101–117, Mar. 2008.
- [6].H. Cao and A. C. Kot, "Accurate detection of demosaicing regularity for digital image forensics," IEEE Trans. Inf. Forensics Security, vol. 4, no. 4, pp. 899–910, Dec. 2009.
- [7].C. Chen and M. C. Stamm, "Camera model identification framework using an ensemble of demosaicing features," in Proc. IEEE Int. Workshop Information Forensics and Security, 2015, Rome, Italy, pp. 1–6

- [8].T. Filler, J. Fridrich, and M. Goljan, "Using sensor pattern noise for camera model identification," in Proc. IEEE Int. Conf. Image Processing, 2008, San Diego, CA, pp. 1296–1299.
- [9]. T. H. Thai, R. Cogranne, and F. Retraint, "Camera model identification based on the heteroscedastic noise model," IEEE Trans. Image Process., vol. 23, no. 1, pp. 250–263, Jan. 2014.
- [10]. E. Kee, M. K. Johnson, and H. Farid, "Digital image authentication from JPEG headers," IEEE Trans. Inf. Forensics Security, vol. 6, no. 3, pp. 1066–1075, Sept. 2011.
- [11]. F. Marra, G. Poggi, C. Sansone, and L. Verdoliva, "Evaluation of residual-based local features for camera model identification," in Proc. Int. Conf. Image Analysis and Processing, 2015, pp. 11–18.
- [12]. M. Kharrazi, H. T. Sencar, and N. Memon, "Blind source camera identification," in Proc. Int. Conf. Image Processing (ICIP), 2004, pp. 709–712.
- [13]. C. C. Loy, T. Xiang and S. Gong, "Multi-camera activity correlation analysis", Proc. IEEE Conf. Comput. Vis. Pattern Recognit., pp. 19881995, Jun. 2009.
- [14]. X. Wang, "Intelligent multi-camera video surveillance: A review" in Pattern Recognit. Lett., vol. 34, pp. 3-19, 2013, [online] Available: http://www.sciencedirect.com/science/article/pii/S016786551200219X.
- [15]. N. Gheissari, T. B. Sebastian and R. Hartley, "Person reidentification using spatiotemporal appearance", Proc. IEEE Conf. Comput. Vis. Pattern Recognit. (CVPR), vol. 2, pp. 1528-1535, Jun. 2006.
- [16]. M. Farenzena, L. Bazzani, A. Perina, V. Murino and M. Cristani, "Person reidentification by symmetry-driven accumulation of local features", Proc. IEEE Conf. Comput. Vis. Pattern Recognit., pp. 2360-2367, Jun. 2010.

- [17]. D. S. Cheng, M. Cristani, M. Stoppa, L. Bazzani and V. Murino, "Custom pictorial structures for re-identification", Proc. Brit. Mach. Vis. Conf., pp. 1-11, Sep. 2011.
- [18]. S. Liao, Y. Hu, X. Zhu and S. Z. Li, "Person re-identification by local maximal occurrence representation and metric learning", Proc. IEEE Conf. Comput. Vis. Pattern Recognit., pp. 2197-2206, Jun. 2015.
- [19]. S. Tan, F. Zheng and L. Shao, "Dense invariant feature based support vector ranking for person re-identification", Proc. IEEE Global Conf. Signal Inf. Process., pp. 687-691, Dec. 2015.
- [20]. T. Matsukawa and E. Suzuki, "Person re-identification using CNN features learned from combination of attributes", Proc. Int. Conf. Pattern Recognit., pp. 2428-2433, Dec. 2016.
- [21]. L. Wu, C. Shen and A. van den Hengel, "PersonNet: Person re-identification with deep convolutional neural networks", arXiv:1601.07255, 2016, [online] Available: https://arxiv.org/abs/1601.07255.
- [22]. R. Varior, M. Haloi and G. Wang, "Gated Siamese convolutional neural network architecture for human re-identification", Proc. Eur. Conf. Comput. Vis., pp. 791-808, Sep. 2016.
- [23]. Zhang et al., "AlignedReID: Surpassing human-level performance in person reidentification", arXiv:1711.08184, 2017, [online] Available: https://arxiv.org/abs/1711.08184.
- [24]. Sun, L. Zheng, W. Deng and S. Wang, "SVDNet for pedestrian retrieval", Proc. IEEE Int. Conf. Comput. Vis., pp. 3800-3808, Oct. 2017.
- [25]. Lin, L. Ren, J. Lu, J. Feng and J. Zhou, "Consistent-aware deep learning for person re-identification in a camera network", Proc. IEEE Conf. Comput. Vis. Pattern Recognit., pp. 3396-3405, Jul. 2017.

- [26]. Zhang, Q. Zhang, X. Wei, Y. Zhang and Y. Xia, "Person reidentification with triplet focal loss", IEEE Access, vol. 6, pp. 7809278099, 2018.
- [27]. Hu, S. Liao, Z. Lei, D. Yi and S. Z. Li, "Exploring structural information and fusing multiple features for person re-identification",
- [28]. Proc. IEEE Conf. Comput. Vis. Pattern Recognit. Workshops, pp. 794799, Jun. 2013.
- [29]. Bazzani, M. Cristani and V. Murino, "Symmetry-driven accumulation of local features for human characterization and re-identification", Comput. Vis. Image Understand., vol. 117, no. 2, pp. 130-144, 2013.
- [30]. Yang, J. Yang, J. Yan, S. Liao, D. Yi and S. Z. Li, "Salient color names for person reidentification", Proc. Eur. Conf. Comput. Vis., pp. 536-551, Sep. 2014.

©"BBIJTM 2022", All Rights ReservedPage 94



"Bodh", BPIT's International Journal of Technology & Management ISSN: 2454-8421, Volume 7, Issue 2, July- Dec 2021, Page 89-104

Importance of values and Ethics for teachers: A reality check for the Contemporary Indian Education Sector

Anuja Thakar<sup>1</sup>

<sup>1</sup>Department of Management Jagannath Institute of Management Studies/Jagannath University, Sector-3, Rohini, New Delhi, India <sup>1</sup>anuja.thakar@jimsindia.org

Abstract- Morals and good values are the building blocks of the character of a person, thereby, of the society as such. Off late, there has been a rampant degradation of ethics and values, pervading all cross sections of the society in terms of age, gender and professions. However, the most potent threat to mankind stems from this degradation pervading the education sector. Teachers being the nurturers of future generations of this country, there is an immediate need to arrest this rot in the system and ensure a value based and ethical education system.

Keywords---Morals, Ethics, Ethics in Education, Ethical Conduct, Value Based Conduct.

#### I. INTRODUCTION

"The function of education is to teach one to think intensively and to think critically. Intelligence plus character – that is the goal of true education."~ Martin Luther King, Jr. (1929-1968) [1]. Morals and values, since time immemorial, have been recognised as the bedrock of human character [2]. History has always regarded us as supreme, the men and women of honour and character. There exist, in history as well as in mythology the world over, numerous counts of persons of virtue giving up everything, including – in some cases – even their lives, to honour their word. That, alas, was history! Those were the times when even the wars were fought with respect and honour for one's enemies, in an ethical manner. It was against the warriors' creed to attack an unarmed opponent or to attack an unprepared enemy from behind. Such morality was founded upon the education of the young amidst an atmosphere of ethics and sound morals.

Today, the world is seeing a constant erosion of morals and values in a never-ending quest for material gains and the greed for power. Surprisingly, while almost every sane mind does recognise this exponential degradation of values, it is also their belief, more or less, that very little can be done to arrest this downfall. While the degradation of morality and values is evident across all sections of the society and assuming monumental proportions, no other field is affected more seriously by this problem as the education sector. The most potent threat to our future stems from the decline of ethical and value based conduct among the teacher, because teachers are the nurturers of future generations. Teaching has always been regarded as the noblest and, arguably, the most challenging profession. It is expected of a teacher to foster in the minds of their pupils the belief in leading an ethical life, following good morals and values. In the recent years, owing to a number of incidents that have taken place, it has oft been commented that the morals, values and ethics in the profession of teaching have also severely eroded. There is an immediate requirement for introspection, to ascertain the extent of evacuation of morality and values from the education sector and also to ensure that the changes in the society notwithstanding, the education sector must be insulated from this erosion of morals and values.

#### II. AIM

The aim of this paper is to examine the relevance of morals and values for the higher education sector, highlight their prevailing state in India and to suggest some measures to ensure moral and ethical conduct by the *teachers and students*. "Leadership is a potent combination of strategy and character, but if you must be without one, be without the strategy[3]" ~ General Norman Schwarzkopf ethics originated from a value based upbringing.

# III. MORALS, VALUES AND THE TEACHER

# A. Morals, Values and Ethics

Character is the most important asset for a man. Nothing describes the importance of character for a human being better than the above mentioned statement by General Norman

H. Schwarzkopf, who led the US military in the Gulf War of 1991. He emphasized that character was more important for warriors, than even military strategy. This also holds true for every other profession. Before we analyze the importance of these virtues to a teacher, it would be prudent to understand what morals, values and ethics actually mean. These words are often used as synonyms for each other. Probably, the dictionary meanings of each of these will amplify the difference and significance of these words. The Cambridge Advanced Learner's Dictionary defines these words as follows (bolds / italics have been added by the author for emphasis on key aspects):-

**Moral [4-5]**: Relating to the standards of good or bad behaviour, fairness, honesty, etc, which each person believes in, rather than to laws. Behaving in ways is considered by most people to be correct and honest. It is a standard for good or bad character and behaviour.

**Values** [6-7]: The beliefs people have, about what is right or wrong and what is most important in life, which controls their behaviour.

**Ethics** [8-9]: A system of accepted beliefs, which controls behaviour, especially such a system based on morals. The study of what is morally right and what is not.

It is evident from the aforesaid dictionary meanings, that while morals and values are largely personal traits of an individual, ethics are more related to group, society or organizational aspects. Ethics enable an action to be considered in a perspective, within the framework of an organization or a society, and then arrive at a conclusion of that action being good or bad. For instance, in certain societies polygamy is a taboo while in another society it is an accepted norm. This is based on the ethics of those respective societies. The profession of teaching, more than any other, ought to be founded on impeccable ethics, as it shapes the future generations of the planet.

Teacher is a Paragon of Virtue. Almost every young mind casts an image in their minds and aspires to be like either their parents, or their teachers. The teacher, as a 'role-model', is expected to set an example for the pupils to follow. However, the young boys and girls carry out a detailed analysis of their teachers and only when found worthy of it do they aspire to emulate them [10]. Besides the pre-requisites of good professional knowledge and acumen, a

teacher ought to be a person of strong character, morally upright and honour-bound. Honesty and integrity are counted among the core values of a teacher, besides the qualities of selflessness and loyalty to the profession. Innumerable accounts of personal sacrifice for each other, from gurus and their disciples, adorn the historical texts. Swami Vivekananda paid the most befitting tribute to his teacher, Swami Ramakrishna Paramhansa, when he said, "I am what I am, and what I am is always due to him; whatever in me or in my words is good and true and eternal came to me from his mouth, his heart, his soul. Sri Ramakrishna is the spring of this phase of the earth's religious life, of its impulses and activities. "If I can show the world one glimpse of my Master, I shall not have lived in vain" [11].

A teacher harnesses tremendous power in himself, bestowed upon him by his ability to nurture the thoughts and influence the minds of his pupils. But that power is accorded to him along with an unwritten undertaking of very onerous responsibility that he would do so only in extreme circumstances, when no other recourse is available. Any misuse or abuse of that power would lead to shattering of the faith of several generations, which rally behind their educators.

# IV. MORALS, VALUES AND ETHICS IN INDIAN EDUCATION SECTOR – A REALITY CHECK

Over the recent few years, especially the last three decades, there have been numerous reasons cited for the rapid decline in moral standards and ethics in Indian society as a whole. Different observers have analyzed the causes for this decline in diverse manners. While some have attributed it to being a fall-out of westernization of Indian culture, others have pinned the blame on immoral conduct of political leadership while some others have done so upon the rampant corruption among government servants and service providing agencies, particularly so the over-ambitious and overtly career conscious leadership in public and private sector enterprises. All of these may, however, not be the most comprehensive and fair assessments of the problem. Among a few of the important questions that need answering is, firstly, "Have the morals and values undergone degradation in the Indian education sector?" Secondly, "If yes, who or what is responsible for it?" and thirdly, most importantly, "What can be done to arrest this decline?"

The answer to the first question is an obvious and resounding "Yes". The morals, values and ethics have indeed taken a beating in the last few decades. The concept of 'Student Before Self' and 'Self-regulated Honour Code', which used to be the hallmark of a teacher have been reduced, by inconsiderate and reckless actions of some individuals, to mere words of insignificance. How else could one explain the several accounts reported over the years, of academic practitioners indulging in immoral actions of monetary gratification being sought for marks and question papers, sexual misconduct, focusing on private tuitions rather than institutional teaching, irregularities in conduct of examinations, etc. ? A perusal of detailed accounts of such actions, as listed above, would reveal that the affliction is not restricted to a particular level and it rather permeates all levels of academic hierarchy.

Therefore, the answer to the second question is that everyone who commits as well as those who ignore/ overlook / condone such actions being carried out, is responsible for it. While some are directly guilty of such behaviour, the balance of us is guilty of condoning the same. The unethical behaviour of a teacher has an all pervading manifestation, across the cross section of his students. Besides, this manifestation takes different forms depending upon the manner in which it affects the students. As has been observed in the recent past, the resultant behaviour may range from small acts of indiscipline of certain individuals, like disobedience of orders, thefts, smoking on campus, use of unfair means in examination halls, eve teasing, etc., to even bigger crimes like molestation, rape or murder, when the moral standards are breached repeatedly and rampantly. The problem, therefore, is more complex than just being considered as erratic or incorrect behaviour of individuals. It must be remembered that the individual under the lens is a teacher and, therefore, upon his actions hinge the dynamics of all the students who look up to their teachers. The ethical dimension, in the field of teaching, therefore, is paramount.

# V. POSSIBLE CAUSES FOR UNETHICAL BEHAVIOUR

While nothing in this world can justify immoral and unethical behaviour, there seem to be some factors which may have a bearing upon such behaviour by some teachers. An observing mind, over the past twenty years has noted the major ones as enumerated:-

- 1. Lack of Emphasis on Morals and Ethics in Training and Early Stages of Life: There is a definite lack of training in ethical behaviour in the formative stages in the country. Training in values and ethics is generally limited to some non-graded value education books. Unless as a child these values and ethics are ingrained into the system, their importance will never be fully realized and generations after generations will see rapid decline in values based living.
- 2. **Result Oriented Approach:** Today's academics, almost like the corporate sector, have got influenced by the target or result oriented approach. The focus is primarily on achieving 'desired results' at all costs, usually by means of stage managing. The focus today is more on results and not on learning or knowledge sharing. This 'results oriented approach' is severely detrimental to the development of value based, ethical actions and a 'humane' psyche, which is more important for a teacher. What one must remember is that the end does not always justify the means.
- 3. Lack of Resources in Institutions: While there is a boom in the education sector, in terms of new institutions being opened up, the quality is taking a severe beating. Some of such institutions lack proper infrastructure for academic pursuits and associated co-curricularactivities. It may be financial or material resources which are at a premium and teachers are asked to be ingenuous and enterprising, to manage things 'by hook or by crook'.
- 4. **Inadequate Avenues for Personal Aspirations:** Every man dreams and aspires to be an achiever. Such avenues in the profession of teaching are limited. This results in discontentment and resentment, prompting them to resort to unethical practices, in

pursuance of their aspirations right from the early days. The means adopted may vary from resorting to promoting students to use unfair means in examinations for better student feedback and visibly better grades in comparison to other teachers, fraternizing with students for popularity, affiliations on regional or religious grounds, to paying bribes for career progression through manipulated and better appraisal reports. Resorting to private tuitions and coaching classes, for monetary gains, and favoring students who subscribe to such services is also very common.

- 5. **Financial Insecurity:** Teaching has long ceased to be a career of choice for the youth. Once regarded as a noble profession, they are now regarded as non-achievers by the more pragmatic and materialistic generation. Those who do join this line usually end up unsatisfied, comparing themselves to the outside world and some of them resort to unethical means in a bid to achieve competing lifestyles with their friends / counterparts in the corporate world.
- 6. **Decline in Respect and Social Status:** Teachers, the shapers of our future generations are not accorded the due respect, financial remuneration and social status which is their rightful due. This gives rise to discontentment and is a cause of severe stress in the teachers in our country.
- 7. **Personal Inadequacies:** There are certain individuals who have flaws and weaknesses in their character. It is possible that despite the stringent selection process, they have been able to make it through and are now a part of the system. Such individuals, if not weeded out of the organization in time, will destroy generations of the future.

- 8. **Inadequate Disposal of Identified Cases of Unethical Conduct:** Not all cases of immoral or unethical behavior in the education system have been satisfactorily disposed. Usually the education sector is being driven by big business entities and cases go unreported or are covered up. Unless the environment realizes that any violations of the moral code, irrespective of the magnitude of such violations, will attract the maximum punitive action, it is unlikely that people with such inclinations will be deterred from indulging into such behaviour.
- 9. Societal Influence: While this reason may initially appear to be as firing the cannon from someone else's shoulder, the influence of the society upon a teacher cannot be ignored. With the details of rampant corruption in the other professions and businesses in the country, there are many people in the teaching profession who feel they are being deprived of this share. At times they feel furious because it is from the taxes they have paid that some cronies are enjoying a lavish life. This resentment and the feeling of deprivation also may lead some people down a path of unethical lifestyle.

# VI. MEASURES TO PROMOTE ETHICAL AND VALUE BASED CONDUCT

Some steps that can be taken to provide an ethical environment in academic institutions and promote value based conduct are as follows:-

- 1. **Stringent and Uncompromising Selection Norms**: The selection process for teachers has to be very rigorous and may be it needs to take a lead from the services selection board (SSB) for military officers, which is regarded as one of the most stringent and nearly fool-proof system for selection of military officer trainees. Proper psycho-analysis of prospective teachers is very important, given the influence a teacher casts on the future generations. It is better to have a deficiency in numbers, rather than have deficiency in moral and ethical standards of teachers.
- 2. Ethics and Values in Training Curricula: There can be no denying the fact that the foundation of a teacher must be based on strong values and ethics. The blasé and nonchalant approach with regard to these issues has to cease right away. Besides the subjects being included in greater details in the curricula, adequate emphasis must also be laid on ethical and value based conduct by officer cadets. Strict adherence to Honour Codes must be implemented and harsh action taken against defaulters, to instill the sense of abiding by such conduct through the life. The tendency among some quarters, to overlook infringements or award light punishments, in the name of shortage of teachers, is inappropriate and must be curbed.
- 3. Emphasis on Quality of Knowledge Sharing Rather Than on Result Oriented Approach: A teacher should be judged on the basis of his overall intellect and moral qualities and not merely on the basis of academic results or honours and awards. While professional knowledge and competence are important facets of personality of a teacher, the marks and results may not be the true reflection of his capabilities. The extent of manipulation in marks and grading, to achieve seemingly good results, has

been repeatedly brought to light from different quarters. More attention, therefore, must be paid to a teacher's character and human qualities and, most importantly, on his performance in shaping the character and wholesome positive personality traits in his students.

- 4. **Exemplary Punitive Action:** As far as ethical and moral dimension is concerned, it is vital to note that there is either 'white' or 'black' and there are no 'grey' zones, therein. There cannot be, under any circumstances, partially moral or ethical behaviour. Therefore, exemplary punishments for violations of moral codes must be meted out, to drive home the importance of abiding by the standards. Long drawn trials followed by light punishments or acquittals, gained by manipulation of systems or exploiting of 'loopholes', encourage the fence-sitters also to take recourse to unfair practices. This must be deterred by swift execution of justice and maintaining scrupulous standards in this context.
- 5. Enhancement of Social and Financial Status of Teachers: There is an inescapable need for enhancing the social and financial status of the teachers, to augment the motivational and security needs of the teaching community. Although some corrective measures have already been instituted towards this effect, especially in government aided institutes, there still is a lot of ground to be covered particularly in the private sector, which incidentally dominates the education market in our country. It may easily be argued that comparison with civil services or the corporate sector must not be solicited in view of different service environments and complexities of the job profiles. However, the comparison is inevitable. The disparity in financial

remuneration and social status must be appropriately bridged to meet the aspirations of the teachers and keep them motivated.

- 6. Address Resource Crunch: It would be incorrect to assume that the resource crunch in terms of basic infrastructure, modern teaching aids, proper and contemporary training material, good classrooms, other ancillary facilities, etc., will not affect the psyche of teachers and, consequently, their personal conduct. All these constraints have a profound impact on the job satisfaction of teachers and the absorption quotient of students as well. In addition, the shortages lead to unscrupulous and adhoc practices which tend to get templated and accepted as the way of life. A person may, therefore, not even realize as to when he transgresses the boundaries of ethical conduct and makes a wrong decision.
- 7. Value Based Conduct in Society: This is a vital ingredient to the moral health of the society. A large segment of population today is bereft of moral values at large. In its quest for material needs and the lack of adequate opportunities for all, morality and ethics have been relegated to lowest rungs of priority. This dereliction is evident from the contents of newspapers and electronic media news every day. However, as it is said that even the longest of journeys begin with a small step, it is for each one to at least try and make a beginning. In addition, the Governments must be urged to include these aspects in our formal education at all levels and ensure implementation of honour codes for citizens, backed by equally stringent implementation supported through legal means.
### VII. CONCLUSION

Change will not take place in a day. That, however, does not imply that efforts for change must not be made in full earnest. The society looks upon the teaching fraternity as the cradle of all virtues and values. Numerous surveys by various organizations, across the cross section of population, have repeatedly indicated that the teachers do command significant respect and admiration from the people. It is noteworthy that a survey by UK based Varkey Foundation, of 2018 indicates 54% of Indian parents encourage their children to take up teaching as a profession [12]. In the same survey, India ranked eighth amongst 35 countries, with respect to social status of teachers [13], providing a ray of hope towards this noble profession. It is prudent, therefore, to undertake immediate measures to arrest the moral decline that has lately affected some sections of the teaching community.

#### REFERENCES

- [1].Bass, Randall V. "The purpose of education." In The Educational Forum, vol. 61, no. 2, pp. 128-132. Taylor & Francis Group, 1997.
- [2]. The Structure of Human Values: A Principal Components Analysis of the Rokeach Value Survey (RVS), Debats,
- [3].D.L. &Bartelds, B.F, <u>https://www.rug.nl/research/portal/files/10223324/c5.pdf</u>, 04 Feb 2019.

- [4].Structure of Human Values: Testing the Adequacy of the Rokeach Value Survey, Braithwaite V. A. & Law H. G., Journal of Personality and Social Psychology, 1985, Vol. 49, No. 1,250-263.
- [5].Michelson, Brian M. Character Development of US Army Leaders: A Laissez Faire Approach. Army war college carlisle barracks pa, 2013.
- [6]. Anscombe, G. E. M. (1958). Modern Moral Philosophy. Philosophy, 33(124), 1-19.www.hindawi.com/journals/isrn/2013/496701/
- [7]. Homer, P.M. and Kahle, L.R., 1988. A structural equation test of the value-attitudebehavior hierarchy. Journal of Personality and social Psychology, 54(4), p.638.
- [8].Hosmer, L.T., 1995. Trust: The connecting link between organizational theory and philosophical ethics. Academy of management Review, 20(2), pp.379-403.
- [9]. Declining Ethical Values in Indian Education System, Shelly &Kusum Jain, Journal of Education and Practice, ISSN 2222-1735, Vol 3, No 12, 2012, https://www.iiste.org/Journals/index.php/JEP/article/view/2876/29 02, accessed on 20 Feb 2019.
- [10]. Impact of Value Based Education on Holistic Personality Development and Management of Academic Stress in Higher Education Institutions of India, Singh, Priyanka, Dayalbagh Educational Institute, Agra, 2016, (https://shodhgangotri.inflibnet.ac.in/bitstream/123456789/2531/1/ synopsis.pdf, accessed on 26 Feb 2019).
- [11]. Study of the Ethical Values of College Students (2006), Mercader, Victor, Graduate Theses and Dissertations. http://scholarcommons.usf.edu/etd/2629.
- [12]. Importance of Human Values in the Society, Debbarma, Dr. Mohan, International Journal of English Language, Literature and Humanities, Volume II Issue I, April 2014, ISSN 2321 – 7065.

[13]. Global Teacher Status Index 2018, The Varkey Foundation. All India Survey on Higher Education: Instruction Manual, Ministry of HRD, Department of Higher Education Planning, Monitoring & Statistics Bureau, 2011.



"Bodh", BPIT's International Journal of Technology & Management ISSN: 2454-8421, Volume 7, Issue 2, July- Dec 2021, Page 105-118

# Relevance of Corporate Governance and Profitability in Innovation Policies of Socio-Economic Transformation: A Study in Indian Context

Shamsher Singh<sup>1</sup>, Neelam Turan<sup>2</sup>

<sup>1</sup>Department of Management, Budha College of Higher Education, affiliated to K.U.K Karnal Haryana, India.

<sup>2</sup>Department of Management, Govt. PG College for Women Karnal affiliated to K.U.K, Haryana, India. <sup>1</sup>drshamshersingh1@gmail.com, <sup>2</sup>neelamturan@gmail.com Sciences Department, VIPS-TC College of Engineering, Delhi, India

#### Abstract-

Corporate Governance is the set of policies that are created for deciding a company's performance and dire ction. In recent years, the area of corporate governance has been in limelight and has attracted increased atte ntion of academicians and researchers worldwide due to high

profile scandals and corporate collapses like Enron, WorldCom and Satyam.Further, we find that corporate profitability also has an insignificant positive impact on governance rating of firm.Before looking at the relati onship between corporate governance, firm performance, and economic growth, it is useful to have a framew ork with which to understand how corporate governance can affect firm behavior and economic performance. One of the problems with the current debate on corporate governance is that there are many different, and o ften conflicting, views on the nature and purpose of the firm.

Keywords---Corporate Governance, Economics,

### I. INTRODUCTION

Corporate governance is a multidisciplinary field of study that covers a wide range of disciplines – accounting, consulting, economics, ethics, finance, law, and management. It also includes the relationships among the various stakeholders (e.g. members/shareholders, management and board of directors) involved and the goals for which the corporation is governed. It refers to leadership structure and values that determine corporate direction, ethics and performance. The aim is to align as nearly as possible the interests of individuals, corporations and society. It focuses on how management is committed to sustainability and

corporate responsibility at all levels. The term corporate governance has been used in many different ways and the boundaries of the subject vary widely. In the economics debate concerning the impact of corporate governance on performance, there are basically two different models of the corporation, the shareholder model and the stakeholder model. In its narrowest sense (shareholder model), corporate governance often describes the formal system of accountability of senior management to shareholders. In its widest sense (stakeholder model), corporate governance can be used to describe the network of formal and informal relations involving the corporation.

More recently, the stakeholder approach emphasizes contributions by stakeholders that can contribute to the long term performance of the firm and shareholder value, and the shareholder approach also recognises that business ethics and stakeholder relations can also have an impact on the reputation and long term success of the corporation. Therefore, the difference between these two models is not as stark as it first seems, and it is instead a question of emphasis.

A large number of studies in the past have examined the relationship between corporate governance and corporate financial performance, but the results have been mixed and inconclusive. Eccles et al. (2012) recommended that for sustainability to be embedded in the organizational culture, the governance structure needs to be tailored accordingly. They emphasized on the significance of two key elements of Corporate Governance, i.e. Board of

Directors (BOD) and Executive Compensation, to ensure sustainable growth of the organization. The overall study objective is to formulate a broad planning and development framework setting out guidelines and standards for more effective and comprehensive planning for corporate governance.

The main objectives of the study are as

- To study the present unique economic situation in the Corporate World.
- To study the relationship between corporate governance and corporate profitability.
- To enhance the long-term shareholder value in corporate governance.

# II. LITERATURE REVIEW

This study will further advance corporate governance by helping people discover more about their planning and development. So in the past by the researchers following are the outcomes. Gompers et al. (2003) performed their study using a governance index for 1500 large US firms, and found that the risk-adjusted returns of firms with strong governance were 8.5% higher than firms with poor governance. Brown and Caylor (2004) found a positive association between corporate governance scores (after adjusting for industry effects) and financial performance of firms (based on dividend payout, yield, profitability and shareholder returns). One of the key

findings of Mani and Sreedharan (2004), a study conducted by CRISIL Ratings in Indian context over a 3 years period was that superior governance practices of firms are positively and significantly correlated to market valuation of firms.

Van de Velde et al. (2005) observed that portfolios with above-average governance scores outperformed the portfolios with below-average governance scores. Governance Metrics International and Byun (2006) found that companies with higher governance ratings enjoyed higher profits and returns. Ashaugh-Skaife and Lafond (2006); Derwall and Verwijmeren (2007) analyzed the linkage between Governance Metrics International (GMI) governance ratings and the firm's cost of equity capital. They suggested that firms with higher governance scores demonstrate lower risk to investors, and thus, enjoy lower cost of capital.

Sachs (2007) suggested that investments in highly governed companies significantly outperform the investments made in poorly governed companies. Bala Subramanium et al. (2008) found a positive and statistically significant association between their overall India Corporate Governance Index (ICGI) and Tobin's Q value (used as proxy for market value of firm) and further this association was stronger for more profitable firms and firms with higher

growth prospects. Selvaggi and Upton (2008) commented on the direction of causality and stated that good governance causes good firm performance, rather than vice versa.

However, Mukherjee and Ghosh (2004) portrayed a dismaying scenario and concluded that corporate governance in India is still in a young and developing stage and that the investment decisions of Indian investors are volatile, not based on governance practices of firms. Chidambaram et al. (2006) found no significant performance difference between firms with good governance changes and firms with bad governance changes and thus, rejected the hypothesis that better governance leads to better financial performance. This result is consistent with those of Core et al. (2006); and Statman and Gluskhov (2009), who found no significant relationship between corporate governance and firm performance. Azim (2012) observed that different governance elements have varying impact on corporate performance and profitability.

Maria Maher and Rhomas Andersson (2016) Corporate governance: effects on firm performance and economic growth Corporate governance has traditionally been associated with the "principal-agent" or "agency" problem. A "principal-agent" relationship arises when the person who owns a firm is not the same as the person who manages or controls it.

The purpose of this study is to present a review of the literature on two lines of research, corporate governance and innovation, explaining how different internal corporate governance mechanisms may be determinants of business innovation. It explores the theoretical background and the empirical evidence regarding the influence of both ownership structure and the board of directors on company innovation. Then, conclusions are drawn and possible future research lines are presented. No consensus was observed regarding the relation between corporate governance and innovation, with both positive and negative arguments being found, and with empirical evidence not always pointing in the same direction. Thus, new studies trying to clarify this relationship are needed. Over recent years, interest has grown in the influence of governance mechanisms on innovation decisions taken by the management.

Innovation efforts and results depend on factors that are influenced by corporate governance, such as ownership structure or the functioning of the board of directors. This paper presents an updated state-of-the-art in this field proposing future lines for empirical, however, states that the relation between corporate ownership and innovation activity may work both ways. He argues that ownership structure affects R&D expenditure, which affects company value, which

in turn affects ownership structure. Other studies do not analyze a direct relation between ownership concentration and innovation but consider that the former may moderate another existing relation. For example, find that cash flow control by owners positively moderates the relation between internationalization and innovation in the company, while the entrenchment effect that arises from the diverging interests between control and cash flow rights may negatively moderate this relation. Other studies propose that R&D expenditure, Ownership concentration and innovations. The prior literature includes arguments supporting both a positive and a negative relation between ownership concentration and R&D activities. An initial argument on a negative relation lies in greater risk aversion. Ownership concentration and combined ownership and management may reduce the pressure that external investors or other supervisors exert on managers in their control of financial statements, information transparency and strategic renewal.

## III. RESEARCH METHODOLOGY

Statistical tools such as descriptive statistics and regression have been applied using Microsoft Excel to study the relationship between corporate governance and corporate profitability using

secondary data. The average data over a period of three years from FY 2016-17to FY 2020-21has been used to enable cross-sectional analysis.

## IV. DATA ANALYSIS AND INTERPRETATION

# Table. 1: Relationship between Corporate Governance and Corporate Financial

S. No.	Study	Relationship
1.	Gompers et al. (2003)	Positive, but not significant
2.	Brown and Caylor (2004)	Positive
3.	Mani and Sreedharan (2004)	Positive and Significant over 3 years period
4.	Mukherjee and Ghosh (2004)	Not Significant
5.	Van de Velde et al. (2005)	Positive, but not significant
6.	Chidambaram et al. (2006)	Not Significant
7.	Core et al. (2006)	Not Significant
8.	Governance Metrics	Positive
	International and Byun (2006)	

Performance

9.	Sachs (2007)	Positive and Significant
10.	Balasubramanium et al. (2008)	Positive and Significant
11.	Selvaggi and Upton (2008)	Positive and emphasizes one way causality
12.	Statman and Gluskhov (2009)	Not Significant
13.	Eisenhofer (2010)	Positive
14.	Pande (2011)	Mixed and Inconclusive
15.	Azim (2012)	Mixed results
16.	Aggarwal (2013)	Positive and Significant over 2 years period
17.	Dr. Shamsher Singh	Positive and Significant
18.	Dr. Neelam Rani	Mixed and Inconclusive

The above table shows mixed and inconclusive results on the relationship between corporate governance and financial performance. Thus, we empirically test this relationship in the given paper in an Indian context. On the one hand, although the legal and regulatory environment affects corporate governance, it is also the case that legal rules and regulations are also, in part, the outcome of different corporate governance systems. For example, systems with dispersed ownership may have a stronger need for regulations that protect shareholder rights. For example, as ownership structure in the US has become more dispersed, the legislative

environment has adapted to the particular needs arising from dispersed ownership. And many European countries are adapting their legislative environments, in particular the strengthening of minority shareholder protection, in response to abuses by controlling shareholders that can arise in their systems of corporate governance. Thus, there is a positive correlation between corporate governance and corporate profitability. Overall, we may conclude that corporate governance has positive but not significant impact on corporate profitability. Hence, we do not reject the first null hypothesis (Ho1).

On the basis of theory and literature review and keeping in view the research objectives, we have formulated the following two hypotheses:

Ho1: Governance rating of a company has no impact on its profitability.

Ha1: Governance rating of a company has an impact on its profitability.

Ho2: Corporate profitability has no impact on the governance rating of a company.

Ha2: Corporate profitability has an impact on the governance rating of a company.

# V. FINDINGS AND CONCLUSION

On an analysis and evaluations of the data, the following findings were found:

1) There is a positive correlation between corporate governance and corporate profitability.

- 2) Governance rating has a positive impact on corporate profitability.
- 3) Governance rating of a company has a significant impact on Return on Sale.
- 4) Corporate governance has a positive but not significant impact on corporate profitability.

Thus, we may conclude that corporate profitability has no significant impact on the governance rating of a company. Hence, we do not reject the second null hypothesis (Ho2).

## **Recommendations:**

The following are the recommendations which should be addressed by the future researchers:

- 1) Limited sample size used in this study so large sample size should be incorporated. Short frame of research covered so frame of research should be large.
- 2) Short frame of research covered so frame of research should be largely manipulated.
- 3) Market-based measures such as share prices, stock returns, market value of firms, etc have been ignored. So this hypothetical value should be highlighted.
- 4) Various control variables (e.g. firm's age, growth, leverage, risk, R&D, industry, etc.) have not been incorporated in our analysis so these must be considered in future research.
- 5) Comprehensive measures should be considered in corporate governance because corporate governance is a vast concept.

# REFERENCES

- [1]. Auerbach, A. and D. Reishus (1988), "Taxes and Merger Decision", in J. Coffee and L. Lowenstein (eds.), Knights, Raiders, and Targets, Oxford University Press, Oxford.
- [2]. Aoki, M., B. Gustaffson and O. Williamson (1990), The Firm as a Nexus of Treaties, Sage Publications, London.
- [3]. Aghion, P. and J. Bolton (1992), "An incomplete contracts approach to financial contracting", Review of Economic Studies, 59, pp. 473-494.
- [4]. Aoki, M. (1994), "The Japanese firm as a system of attributes: A survey and research agendas", in M. Aoki and R. Dore (eds.), The Japanese Firm: Sources of Competitive Strength, Oxford University Press, Oxford.
- [5].Becht, M. and E. Bohmer (1999), "Ownership and voting power in Germany", in Barca, F. and M. Becht (eds.), Ownership and Control: A European Perspective, forthcoming.
- [6].Brown, L. D., & Caylor, M. L. (2004). Corporate Governance Study: The Correlation between Corporate Governance and Company Performance. Institutional Shareholder Services (ISS). Retrieved from http://www.stybelpeabody.com/issscoresandshareholdervalue.pdf
- [7].Core, J. E., Guay, W. R., & Rusticus, T. O. (2006). Does weak governance cause weak stock returns? An examination of firm operating performance and investors' expectations. The Journal of Finance, 61(2), 655-687
- [8].Eccles, R. G., Ioannou, I., &Serafeim, G. (2012). The impact of a corporate culture of sustainability on corporate behavior and performance (No. w17950). National Bureau of Economic Research.
- [9].Maria Maher and Thomas Andersson(2016) corporate governance: effects on firm performance and economic growth
- [10]. Fama, E.F. and Jensen, M.C. (1983), "Separation of ownership and control", Journal of Law and Economics, Vol. 26 No. 2, pp. 301-325.

[11]. Naldi, Lucia, Mattias Nordqvist, Karin Sjöberg, and Johan Wiklund. "Entrepreneurial orientation, risk taking, and performance in family firms." Family business review 20, no. 1 (2007): 33-47.